Wintertime Orographic Cloud Seeding—A Review

ROBERT M. RAUBER,^a BART GEERTS,^b LULIN XUE,^c JEFFREY FRENCH,^b KATJA FRIEDRICH,^d ROY M. RASMUSSEN,^c SARAH A. TESSENDORF,^c DEREK R. BLESTRUD,^e MELVIN L. KUNKEL,^e AND SHAUN PARKINSON^e

^a Department of Atmospheric Sciences, University of Illinois at Urbana–Champaign, Urbana, Illinois ^b Department of Atmospheric Science, University of Wyoming, Laramie, Wyoming

^c Research Applications Laboratory, National Center for Atmospheric Research, Boulder, Colorado

^d Department of Atmospheric and Oceanic Sciences, University of Colorado Boulder, Boulder, Colorado

^e Idaho Power Company, Boise, Idaho

(Manuscript received 20 December 2018, in final form 2 August 2019)

ABSTRACT

This paper reviews research conducted over the last six decades to understand and quantify the efficacy of wintertime orographic cloud seeding to increase winter snowpack and water supplies within a mountain basin. The fundamental hypothesis underlying cloud seeding as a method to enhance precipitation from wintertime orographic cloud systems is that a cloud's natural precipitation efficiency can be enhanced by converting supercooled water to ice upstream and over a mountain range in such a manner that newly created ice particles can grow and fall to the ground as additional snow on a specified target area. The review summarizes the results of physical, statistical, and modeling studies aimed at evaluating this underlying hypothesis, with a focus on results from more recent experiments that take advantage of modern instrumentation and advanced computation capabilities. Recent advances in assessment and operations are also reviewed, and recommendations for future experiments, based on the successes and failures of experiments of the past, are given.

1. Introduction

The U.S. population more than doubled from 1950 to 2010 and shifted from rural to urban areas (U.S. Census Bureau 2010). Southern and western states experienced the greatest population increase, resulting in concurrent expansion of public water supply systems. In response to increased demands and limits on water supplies, western communities have sought additional water sources through technologies such as cloud seeding, and/or have instituted water-conservation measures to preserve existing supply (Kenny et al. 2009). Water will become an increasingly scarce resource as populations continue to grow and changes in climate over the coming decades threaten the water volume of snow reservoirs in the western mountains (Mote et al. 2005; Rasmussen et al. 2011).

Across the western United States during winter, precipitation falls as snow over higher elevations along coastal ranges, and at nearly all elevations over interior mountain ranges. The ensuing snowmelt in spring and summer then

Corresponding author: Robert M. Rauber, r-rauber@illinois.edu

provides annual water supplies. As early as the 1950s, following the discoveries of Schaefer and Vonnegut concerning cloud seeding (Schaefer 1946; Vonnegut 1947), water resource managers recognized that seeding wintertime orographic cloud systems had the potential to increase water supplies in arid regions. Increasing winter snowpack through seeding was envisioned as a means to enhance the natural snow reservoir that supplies water to drainage basins throughout the melt season. Indeed, the demand for water drove pioneering scientists in the 1950s to develop projects to evaluate the scientific basis for weather modification as a tool to increase water supplies. The early studies of orographic cloud seeding, which progressed to include elaborate field investigations in the 1970s and 1980s, made some progress in understanding the conditions under which cloud seeding could enhance precipitation, but were unable to clearly establish the magnitude of that enhancement. In its 2003 report, the National Research Council (NRC 2003) stated that "there still is no convincing scientific proof of the efficacy of intentional weather modification efforts." Despite this uncertainty, operational winter orographic weather modification has continued in most western states of the United States

DOI: 10.1175/JAMC-D-18-0341.1

^{© 2019} American Meteorological Society. For information regarding reuse of this content and general copyright information, consult the AMS Copyright Policy (www.ametsoc.org/PUBSReuseLicenses).



FIG. 1. Operational cloud-seeding project target areas for enhancement of winter snowpack in the mountains of the western United States in 2015 (colored green). No projects occurred in the states that are colored blue. (Source: North American Weather Modification Council.)

(Fig. 1) and in other arid regions, a direct result of the increasing need for water, and the large potential cost benefit for production of additional water by cloud seeding.

The fundamental hypothesis underlying cloud seeding as a method to enhance precipitation from wintertime orographic cloud systems is that a cloud's natural precipitation efficiency can be enhanced by converting supercooled water to ice upstream and over a mountain range in such a manner that newly created ice particles can grow and fall to the ground as additional snow on a specified target area. Orographic clouds, in this context, refers to cloud systems over mountain ranges, whether isolated or components of frontal systems within extratropical cyclones. This static-seeding hypothesis has its roots in the physical principle that the equilibrium vapor pressure with respect to ice is less than that with respect to liquid water at the same subfreezing temperature. Thus, at temperatures below 0°C, a water-saturated cloud (relative humidity with respect to water $RH_w = 100\%$) will be supersaturated with respect to ice at a rate of about 1% per degree Celsius of supercooling (Pruppacher and Klett 2010). The consequence is that in an initially watersaturated cloud containing supercooled water, ice particles grow rapidly to precipitation size, whereas small and nonprecipitating supercooled cloud droplets are either growing in an updraft or evaporating, providing moisture for ice growth. Alfred Wegener first proposed

this diffusional growth process for liquid-saturated clouds (Wegener 1911). It was later explained theoretically and demonstrated experimentally by T. Bergeron and W. Findeisen (Bergeron 1935; Findeisen 1938).

Scientific evaluation of the static-seeding hypothesis has been attempted over the last half century in a number of projects (Fig. 2) using observational processoriented studies to understand natural cloud structure and effects of cloud seeding, statistical comparisons of surface precipitation between treated and untreated events, and numerical models to simulate both natural and seeded clouds. Past reviews at least partially focused on orographic weather systems and/or weather modification research to modify those cloud systems include those of Smith (1979), Elliott (1986), Rangno (1986), Reynolds (1988), Orville (1996), Bruintjes (1999), Long (2001), Garstang et al. (2005), Huggins (2008), Tessendorf et al. (2015), Reynolds (2015), Gultepe (2015), and Haupt et al. (2019). This paper provides a systematic assessment of our current understanding of the effectiveness of wintertime cloud seeding to enhance mountain snowpack, drawing extensively on results from recent studies not available to authors of past reviews, with a focus on results from new and advanced instrumentation, improved understanding of cloud dynamical and microphysical processes, and more sophisticated numerical modeling technologies. We note that many instrumentation platforms have been deployed in efforts to evaluate cloud seeding. It is beyond the scope of this paper to review their uses, accuracy, and effectiveness. The reader is referred to the American Meteorological Society monograph Ice Formation and Evolution in Clouds and Precipitation: Measurement and Modeling Challenges (McFarquhar et al. 2017) for a concise summary of issues related to measurements of ice and snow in clouds and to articles by Rasmussen et al. (2012), Gultepe et al. (2016) and Kochendorfer et al. (2017) for surface measurements of snow.

This paper reviews research related to glaciogenic seeding of winter orographic clouds using silver iodide (AgI), dispersed pyrotechnically from the ground or aircraft, specifically within clouds with cloud-top temperatures typically colder than from -6° to -8° C. Three different approaches to orographic cloud-seeding evaluations are discussed: physical experiments, statistical evaluations, and modeling studies. Section 2 reviews physical evaluations of the hypothesis, summarizing what has been learned about the thermal, kinematic, and microphysical structure of natural winter mountain cloud systems, the transport and dispersion of seeding plumes, and the microphysical chain of events that occurs when an orographic cloud is seeded. Section 3 examines statistical evaluations of the hypothesis, focusing



FIG. 2. Locations of major research projects in the western United States that were designed to evaluate the feasibility of orographic enhancement of snowpack through cloud seeding over the period 1960–2019.

on what has been learned by target/control and randomized evaluations of snowpack enhancement. Section 4 reviews modeling evaluations of the impacts of AgI seeding on orographic clouds. Section 5 considers recent advances in assessment and operations. The paper concludes with a summary of the state of the science and a look to the future.

2. Physical evaluations of the underlying hypothesis of orographic cloud seeding

Physical evaluations of the orographic cloud-seeding hypothesis follow four basic thrusts: 1) determining when and where supercooled liquid water (SLW) is present in clouds, 2) documenting natural precipitation processes and determining which orographic clouds are suitable for treatment with AgI, 3) determining conditions under which plumes of ground-released or airbornereleased AgI reach clouds upstream and within target river basins, and 4) documenting the microphysical chain of events following seeding to determine whether it is consistent with the hypothesis and if the fallen snow contributes to snowpack enhancement.

a. Supercooled liquid water distribution

The first component of hypothesis evaluation has been to determine when and where SLW is present in orographic cloud systems. Mountain ranges were targeted in the first place because it was believed (correctly so) that SLW is commonly present in or near orographic updrafts. Understanding the SLW distribution in clouds was the focus of studies using aircraft, microwave radiometers, balloonborne instruments, and surface measurements conducted over California's Sierra Nevada (Reinking 1979; Heggli et al. 1983; Heggli and Rauber 1988; Demoz et al. 1993), Colorado's Park Range (Rauber et al. 1986; Rauber and Grant 1986, 1987), Utah's Wasatch (Hill and Woffinden 1980; Sassen et al. 1986, 1990), Idaho's Payette Mountains (Tessendorf et al. 2019), Washington's Cascades (Hobbs 1975a; Ikeda et al. 2007), Wyoming's Medicine Bow Mountains (Politovich and Vali 1983; Jing and Geerts 2015; Rasmussen et al. 2018), Australia's Great Dividing Range and Snowy Mountains (Long and Carter 1996), and the central mountains of Japan (Kusunoki et al. 2004, 2005) [see also Reynolds and Dennis (1986), Rauber (1987), Hill (1980), Sassen and Zhao (1993), and Long and Huggins (1992) for additional information on field campaigns]. These studies, which produced consistent results, showed that SLW is most often found in clouds with sufficiently strong updrafts so that the condensate supply rate could be expected to exceed the diffusional growth rate of ice. Detailed measurements with various observation platforms, including airborne profiling W-, K-, and X-band Doppler radars, indicate that such updrafts can be found 1) along and over steep mountain slopes (Rauber et al. 1986); 2) in embedded convection, when it exists (Ikeda et al. 2007); 3) in cloud-top generating cells, particularly for cloud tops warmer than $\sim -25^{\circ}$ C (Rauber and Tokay 1991; Morrison et al. 2013; Keeler et al. 2016a,b, 2017); 4) in updrafts associated with mountain-induced gravity waves (Reinking et al. 2000; Bruintjes et al. 1994); and 5) in turbulent eddies near the mountain surface induced by local terrain (Lee 1988; Geerts et al. 2011; Chu et al. 2018) or within Kelvin-Helmholtz billows and turbulence in shear zones upwind (Houze and Medina 2005; Medina and Houze 2015) or downwind (Geerts and Miao 2010; Barnes et al. 2018; Conrick et al. 2018) of terrain. Supercooled water has also been found 6) in shallow convection (coupled with surface) with cloud tops warmer than about -25° C (Heggli et al. 1983; Heggli and Rauber 1988); 7) in more laminar orographic clouds where cloud-top temperatures are greater than -15°C and ice processes are inefficient (Tessendorf et al. 2017); and 8) in orographic clouds with bases below melting level and terrain-forced ascent of cloud water through the 0°C level (Marwitz 1987; Rauber 1992; Ikeda et al. 2007).

b. Natural precipitation processes

A second component of hypothesis evaluation has been to document natural precipitation processes and to determine which orographic clouds, if any, are suitable for treatment with AgI to enhance precipitation. Again, numerous field campaigns have been conducted across mountain regions to study natural precipitation processes and evaluate the seedability of orographic clouds (e.g., Hobbs 1975a; Cooper and Saunders 1980; Cooper and Vali 1981; Marwitz 1987; Rauber 1987; Uttal et al. 1988; Sassen et al. 1990; Rauber 1992; Long and Carter 1996; Geerts et al. 2010; Ritzman et al. 2015; Rasmussen et al. 2018; Tessendorf et al. 2019). Together, these studies show that the microphysics of mountain cloud systems evolve in close relationship to their mesoscale dynamical structure, which in turn is associated with the approach and passage of surface and upper-tropospheric fronts and jet stream-related circulations. Deep orographic cloud systems, which often occur prior to frontal passage, are typically characterized by ice nucleation, primarily but not exclusively near cloud top within cloud-top generating cells or gravity waves, followed by diffusional growth and aggregation of ice particles during fallout to the surface. Studies in the Cascades of western North America (Hobbs 1975a; Stoelinga et al. 2003), northern Colorado Rockies (Rauber 1987), San Juan Mountains of Colorado (Cooper and Saunders 1980), and Australia's southern mountains (Long and Carter 1996) all support this basic microphysical evolution. Riming, when it occurs, is limited to areas where supercooled droplets are present, especially in strong updrafts. Shallow orographic clouds with cloud-top temperatures greater than -15°C can produce primary ice, but often in concentrations insufficient to consume supercooled water in the cloud, unless ice crystals are introduced from blowing snow near the ground (Geerts et al. 2011; Vali et al. 2012; Geerts et al. 2015b), or if cloud-top generating cells are present that produce plumes of ice particles (Plummer et al. 2014; Rosenow et al. 2014; Kumjian et al. 2014). The Hallett-Mossop ice multiplication mechanism (Hallett and Mossop 1974) can be active in more maritime clouds, such as over the Sierra Nevada and Cascades (Marwitz 1987; Rauber 1992), but it is rare inland because the cloud base is often too cold and few large droplets exist (Cooper and Saunders 1980; Rauber 1987). Shallow orographic clouds sometimes contain embedded convection, both over coastal ranges (e.g., Hobbs 1975a; Heggli et al. 1983; Rauber and Grant 1987; Rauber 1987; Ikeda et al. 2007) and in the interior (e.g., Kumjian et al. 2014; Geerts et al. 2015a). Some days these convective orographic clouds and precipitation can persist for hours as suggested by reflectivity profiles from airborne W-band cloud radar collected during the AgI Seeding Cloud Impact Investigation (ASCII) field program in Wyoming (Geerts et al. 2011; Geerts et al. 2013; Pokharel et al. 2014a; Chu et al. 2014) and the Seeded and Natural Orographic Wintertime clouds: The Idaho Experiment (SNOWIE) field campaign in southwestern Idaho (French et al. 2018; Tessendorf et al. 2019). Conditions on convective days are often marked by weak static stability (i.e., moist Brunt-Väisälä frequency is small) and a layer of potential instability (Kumjian et al. 2014; Geerts et al. 2015a; Pokharel and Geerts 2016).

c. Transport and dispersion of aerosol plumes in complex terrain

A third component of hypothesis evaluation has been to determine if ground-released and/or airborne-released AgI plumes reach targeted mountain regions and sufficient altitudes above ground level, and what dynamical processes are important for the spread of the plumes. Two general frameworks, Lagrangian and Eulerian, are commonly used to simulate particle transport and dispersion in the atmosphere. Lagrangian particle trajectory and dispersion models are commonly used by the air quality modeling community. In the weather modification community, trajectory analysis (e.g., Hobbs 1975b; Holroyd et al. 1988) or simple trajectory models (e.g., Rauber et al. 1988) have been applied for cloud-seeding applications. More recently, sophisticated parcel trajectory models such as the Hybrid Single-Particle Lagrangian Integrated Trajectory model (HYSPLIT; Stein et al. 2015) and plume dispersion models such as the Second-Order Closure Integrated Puff (SCIPUFF) model (Sykes and Gabruk 1997) have been used to investigate the dispersion of AgI emitted from ground-based generators in mountainous areas (Fig. 3). Their accuracy depends on the accuracy and resolution of the 4D numerical model driver dataset and the assumptions on instability and turbulence features applied by these models.

Simulations of AgI dispersion from ground generators have been difficult to validate with detailed observations because of safety considerations in flying research aircraft within plumes close to the surface near mountain peaks; airborne measurements within seeding plumes from ground generators have only been made under clear skies (Boe et al. 2014). Instead, the snowpack in the seeded region has been sampled during or after seeded events to detect anomalous concentrations of silver in snow (e.g., Fisher et al. 2016, 2018) or the ratio of silver to a nonnucleating, naturally covarying aerosol tracer, typically indium oxide (In_2O_3) . Warburton et al. (1995) and Manton and Warren (2011), for example, examined the ratio of silver to Indium in snowfall during ground generator seeding events in the California Sierra Nevada and Australia's Snowy Range, respectively. The ratios of silver to indium showed clear evidence that AgI, acting as an ice nucleant, was selectively incorporated into ice crystals and deposited as snow on the mountains.

Early modeling studies by Super (1974) found that the plumes from AgI released from ground-based generators in a cloud-free and stable atmosphere were confined to the lowest 500 m above the terrain over the Bridger Range in Montana. Holroyd et al. (1988) showed that ground-released plumes over Colorado's Grand Mesa ascended upward at $\sim 2 \text{ m s}^{-1}$ in cloudy environments, again confined within 500 m above the terrain. More detailed modeling studies of plume dispersion by Bruintjes et al. (1995) showed plumes remained lower than 800 m above maximum terrain height when released from an upwind ridge. This height corresponds with the typical

turbulent planetary boundary layer (PBL) depth over a mountain in winter storms (Geerts et al. 2011).

Bruintjes et al. (1995) was the first study in the weather modification community that compared simulated gaseous tracer concentrations [sulfur hexafluoride (SF_6)] using a 3D Eulerian model with airborne in situ observations, demonstrating the capability of a model to qualitatively and quantitatively capture the dispersion of AgI in complex terrain. The interaction between the airflow and the topography was identified as the dominant factor in determining the transport and dispersion of the seeding agent. The modeling study suffered from relatively coarse grid spacing (2 km) so that only the largest turbulent eddies were resolved. In more recent years, large-eddy simulation (LES) models have been used to explicitly compute 3D turbulent motions at higher spatial resolutions on the order of 100 m enabling a better estimation of the 3D dispersion of particles. LES can reasonably reproduce the wind field and turbulent kinetic energy in complex terrain regions (Chow et al. 2006; Weigel et al. 2007). Using a 3D Eulerian model that simulates AgI seeding as part of the microphysical parameterization (Xue et al. 2013a), Xue et al. (2014) were able to demonstrate that the simulated dispersion of AgI particles from five ground-based generators matched airborne and ground-based measurements taken by Boe et al. (2014) over the Medicine Bow Mountains within 50% (see Fig. 4 for examples of AgI dispersion over complex terrain). They used LES with a horizontal grid spacing of 100m and a stretched vertical coordinate (averaged grid spacing of 70 m below 2 km AGL) that resolved most energetic turbulent eddies induced by terrain. Based on their study, the terrain-induced wind shear generates 3D turbulent eddies that are responsible for the vertical dispersion of AgI. In cloudy conditions, both the buoyancy associated with cloud instability and shear induced by terrain were found to be responsible for the vertical dispersion of AgI (Xue et al. 2016). Lowerresolution non-LES simulations (on the order of 2-km grid spacing) using PBL schemes strongly underestimated the vertical dispersion of AgI (Xue et al. 2014). These results suggest that an appropriate and/or improved PBL scheme that incorporates effects of terrain-induced turbulence is needed for lower-resolution simulations (on the order of 2-km grid spacing) in complex terrain since it remains currently computationally impractical to run realtime and/or large numbers of simulations in LES mode.

d. Physical observations of microphysical chain of events during seeding

A fourth component of hypothesis evaluation has been to document ice formation and precipitation evolution from the point of releasing AgI from either ground



FIG. 3. Examples of (left) WRF-simulated and (right) HYSPLIT-simulated AgI number concentrations using a logarithmic scale $[\log(m^{-3})]$ at three vertical levels: (a) 2500, (b) 3000, and (c) 3500 m MSL. The ground-based AgI generators are at the southern edge of the plume indicated by the stars on the HYSPLIT maps. The WRF maps include terrain contours (1500 and 2500 m) and state lines for Idaho.

generators or an aircraft to precipitation reaching the ground. The most successful approaches have been to identify radar echoes corresponding in time and space to the calculated position of newly formed ice particles and precipitation based on advection speed of the AgI aerosol following the release of the seeding agent by an aircraft (referred to as seeding lines or seeding signatures), to document airborne microphysical measurements of anomalously high ice particle concentrations within the seeding lines, and to record any precipitation enhancement as ice particle plumes reach the surface. The primary limitation facing researchers has been extracting the signal



FIG. 4. Three-dimensional depictions of the topography of the Medicine Bow Range in Wyoming, AgI aerosol number concentration (>100 L⁻¹ for visible plumes), and wind vectors at three levels [\sim 2800 (yellow), \sim 3600 m (blue), and \sim 4400 (purple) m MSL] at three times separated by 90 min, shown as a (left) bird's-eye view from the south and (right) side views from the southeast. (Adapted from Xue et al. (2014).]

in cloud systems that are characterized by a high degree of natural variability (Gultepe et al. 2014). In general, the best, and clearest seeding signatures have been obtained in shallow orographic clouds containing supercooled water and few ice particles, since these clouds naturally produce weak or no radar echoes, even at short W- and K-band wavelengths, and ice particle concentrations within plumes created by seeding stand out well above background concentrations.

The first physical evidence of a seeding effect in an orographic environment was presented by Hobbs (1975b)

in three case studies of airborne seeding of stratocumulus and cumulus clouds over the Cascades. In each case, enhanced in-cloud ice particle concentrations, transitions in ice particle habits, increases in silver concentrations in surface snow, and increases in snowfall rate were observed in space and time consistent with expectations based on ice particle plume trajectory calculations. Hobbs et al. (1981), Marwitz and Stewart (1981), Prasad et al. (1989), Deshler and Reynolds (1990), and Chu et al. (2017a) reported similar aircraft and/or radar observations. As part of the Sierra Cooperative Pilot 2124

Project, Deshler et al. (1990) reported results from two experiments where in situ cloud microphysics and groundbased remote sensing platforms recorded seeding effects. Flying through areas affected by seeding, the research aircraft observed a high concentration, $50-100 L^{-1}$, of small ice particles and rimed particles commencing 5-10 min after seeding. Otherwise nonechoing cloud regions produced K_u-band radar echoes of 3-10 dBZduring passage of the seed lines. Seeding effects arrived downwind at the surface 35-60 min after seeding, as ice particle concentrations increased, habits changed to rimed particles, and precipitation rates increased by $0.1-1.0 \text{ mm h}^{-1}$.

Results from airborne seeding during the 2017 SNOWIE campaign in Idaho (French et al. 2018; Tessendorf et al. 2019) illustrate remarkably clear radar seeding signatures, appearing as zig-zag patterns of X-band radar reflectivity in low-elevation-angle horizontal scans (Fig. 5d), and as vertical plumes of W-band reflectivity observed by the airborne Wyoming Cloud Radar (WCR) at the expected location of seeding lines in vertical cross sections (Fig. 5c). The zigzag pattern in Fig. 5d resulted from the seeding aircraft flying repeated legs normal to the prevailing wind and the seeding material being transported downwind. The seeding aircraft burned AgI flares behind the aircraft wings resulting in semicontinuous horizontal lines and dropped AgI flares through the clouds resulting in discrete vertically oriented plumes. The initiation and growth of ice particles created by seeding were also documented in situ using aircraft probes on the University of Wyoming King Air aircraft. For the flight leg shown in Fig. 5a, hydrometeor size spectra collected at flight level showed that inside seed plume 1 aggregates and rimed ice particles up to 4 mm in diameter were present in concentrations of $3-20 L^{-1}$ (Fig. 5e black line and Fig. 5f right-hand image) while just upwind of seed plume 1 nearly all particles were liquid and had diameters less than $50 \,\mu m$ (Fig. 5e blue line and Fig. 5f left-hand image). French et al. (2018) show, from a different flight than the one presented here, the microphysical and radar evolution of similar seeded regions over seven flight legs as an ice-crystal plume evolved and the bottom of the plume reached the mountain surface.

Evidence of seeding effects has also been reported for ground-based seeding studies, however these studies have more sources of ambiguity than the results from airborne seeding during SNOWIE. Super and Boe (1988), Super and Heimbach (1988), and Huggins (2007) reported indications of ground-based AgI seeding in aircraft-measured ice particle concentrations and/or in precipitation at the ground in their studies of stable orographic clouds over Colorado's Grand Mesa, Montana's Bridger Range, and Utah's Wasatch Plateau. They observed, with aircraft, changes in cloud structure consistent with seeding, and enhanced precipitation rates (several times as large as outside the plumes but generally light—less than 1 mm h^{-1}) at the ground coincident with measured AgI plumes. However, the natural variability in ice particle concentrations and snowfall were more difficult to distinguish with certainty from changes in these parameters due to the ground-based seeding in these studies.

A similar technique was applied in ASCII to determine the times when ground-based seeding plumes impacted a downwind manned research station, equipped with a variety of probes including a disdrometer, a particle imaging probe, crystal photography, snow gauges, and a profiling K-band radar (Pokharel et al. 2014b, 2017). ASCII also deployed an X-band scanning radar on the ground and the WCR. Flight-level particle sizing and imaging data were collected as well, but except in a few cases, the flight level (chosen to be as low as permissible in cloudy conditions) was too high to sample groundreleased AgI nuclei (Boe et al. 2014; Xue et al. 2014). In four cases with convective clouds in ASCII, the magnitude of the upward Doppler velocities observed by the WCR provided strong evidence that PBL air downstream of the AgI generators was lofted into clouds, and in situ cloud imaging probe data for University of Wyoming King Air (UWKA) interceptions of those clouds reveal more numerous but smaller ice crystals, compared to clouds over similar terrain on the sides of the AgI plumes, as well as compared to similar clouds over the same terrain before seeding started (Fig. 10 in Pokharel et al. 2017).

Most storms sampled in ASCII occurred in unblocked flow, were nonfrontal, and produced light natural snowfall from rather shallow clouds with little SLW contained in small droplets ($<25 \,\mu$ m) (Pokharel and Geerts 2016; Pokharel et al. 2015). While the vertically integrated liquid water path (LWP) was less than 0.4 mm in ASCII clouds, the lack of stratification and the strong winds over the target mountains allowed effective in-cloud AgI mixing (Geerts et al. 2011; Aikins et al. 2016), especially in convective events (Chu et al. 2017b).

Most storms in ASCII were sampled for two hours under natural conditions, prior to a two-hour seeding period, enabling the examination of temporal differences. Comparisons were also made between the target region (within the assumed AgI plumes) and lateral and upstream control regions. A survey of all 25 ASCII cases found that the radar and snow gauge data agreed in that, in most individual cases and on average, the precipitation rate was higher during seeding, at least in terms of a double difference (i.e., where changes in the target region are compared to those in the control



FIG. 5. Example of seeding lines observed during the 2017 SNOWIE campaign. Shown, from a single pass by the Wyoming King Air, are in situ measured (a) hydrometeor concentration for liquid droplets (blue; from Cloud Droplet Probe) and ice particles (red; only particles >50 μ m in diameter, from 2DS probe) and (b) bulk condensed water content for liquid (blue) and ice (red), both from the deep-cone Nevzorov probe. (c) The vertical cross-section of W-band reflectivity during the same pass. The horizontal dotted line is the flight track. The locations of echoes resulting from seeding plumes are highlighted (green line). (d) A time-coincident 0.5° plan-position indicator scan using a ground-based X-band Doppler on Wheels, with the horizontal extent of the echo resulting from the seeding plume also highlighted. Also shown are 2DS probe particle size distributions (e) measured inside the plumes (black) and just upwind of the plumes (blue) and (f) corresponding two-dimensional hydrometeor shadows (left) outside the plumes and (right) inside the plumes.

regions; Pokharel et al. 2017, 2018). In some ASCII cases, especially those with few ice crystals during the untreated period, ice-crystal concentrations substantially increased during the cloud-seeding period (Pokharel et al. 2014a,b, 2015; Jing et al. 2015). The ASCII clouds, seeded from the ground, did not reveal the clear radar seeding signatures that were encountered in SNOWIE clouds (which were seeded from the air); the most apparent seeding impact was in a very shallow capped

cloud that produced negligible natural snowfall (Chu et al. 2017a). This reflectivity increase, seen in WCR transects, was confirmed by an LES model simulation with the Xue et al. (2013a,b) seeding parameterization (see section 5). In Chu et al. (2017a), the simulated reflectivity was calculated for S-band wavelength. No Mie scattering was assumed for the corresponding W-band observations. The purpose was to compare the reflectivity trend before and during the seeding period

Project Name	Location	Years	Type of expt
Climax I; Climax II	Climax, CO	1960-65; 1966-70	Exploratory; confirmatory
Wolf Creek Pass	Wolf Creek Pass, CO	1964–70	Exploratory
Elko, NV	Northeast Nevada Range, NV	1961–67	Exploratory
Colorado River Basin Pilot Project (CRBPP)	San Juan Mountains, CO	1970–75	Exploratory
Bridger Range Experiment	Bridger Range, MT	1970–72	Exploratory
Wyoming Weather Modification Pilot Project (WWMPP)	Medicine Bow/Sierra Madre Ranges, WY	2008–13	Confirmatory
Snowy Precipitation Enhancement Research Project (SPERP1; SPERP2)	Snowy Mountains, Australia	1955–63; 2005–09; 2010–13	Exploratory; confirmatory; confirmatory;

TABLE 1. Summary of the seven randomized seeding experiments studying the effect of winter orographic precipitation enhancement using silver iodide.

under the assumption that the W-band signals respond to the seeding in the same sense as the S-band signals. In Chu et al. (2017a, their Figs. 6 and 8), the >10-dBZ increase of observed reflectivity in the target from noseed period to seed period was captured by the modelderived S-band reflectivity signal changes.

3. Statistical evaluations of the underlying hypothesis of orographic cloud seeding

Statistical evaluations of randomized cloud-seeding experiments are considered *exploratory* if the study considers a number of hypotheses/analyses, typically guided by understanding gained from physical studies. Evaluations are considered *confirmatory* if anticipated effects are stated ahead of time (a priori), and then proven at an acceptable level of statistical significance following the stated design of the experiment (Gabriel 1999; Silverman 2001, 2007). A variety of ratio statistics have been used in the design and evaluation of weather modification experiments and their significance has usually been estimated by rerandomization (Gabriel 1999). A result is deemed statistically significant when it is very unlikely to have occurred, given a null hypothesis that seeding had no effect. In cloud-seeding studies, this level of significance is typically at the 5% level, that is, the probability of an increase in precipitation due to seeding occurring by chance is less than 5%. The term multiplicity refers to the case in which tests are carried out following a failed exploratory or confirmatory experiment, beyond those tests stated a priori, in search of parameters that yield a statistically significant result.

There have been seven randomized scientifically based projects studying physical effects of cloud seeding for the purpose of increasing seasonal mountain snowpack (Table 1) All experiments employed ground-based seeding with AgI. Other randomized projects have been carried out targeting frontal rainbands in winter environments with the goal of increasing rainfall. These include Santa Barbara I and II (Neyman et al. 1960; Bradley et al. 1979), Tasmania I, II, and III (Ryan and King 1997; Morrison et al. 2009), and Israel I and II (Gagin and Neumann 1981; Silverman 2001). Projects targeting frontal rainbands to increase rainfall are not considered in this review.

The Climax and Colorado River basin Pilot Project (CRBPP) experiments had a single targeted area, randomly treated in 24-h seeded or unseeded intervals over the entire winter season (Mielke et al. 1970, 1971, 1981; Chappell et al. 1971; Elliott et al. 1978; Vardiman and Moore 1978; Rottner et al. 1980, Mielke 1995; Gabriel 1995). The Wolf Creek Pass Experiment, by contrast, seeded during three randomly chosen winter seasons, and left three other seasons untreated (Grant and Elliott 1974; Mielke et al. 1977). All three of these experiments were later shown to contain flaws in design (e.g., 24-h treatment units were too long), employ erroneous assumptions (e.g., the 500-hPa temperature was assumed to represent cloud-top temperature), or have errors in data handling that rendered their statistical results unreliable (Hobbs and Rangno 1979; Rangno 1979; Rangno and Hobbs 1980a,b, 1981, 1987, 1993; Gabriel 1995). The Elko experiment was briefly mentioned in Grant and Elliott (1974). Grant and Elliott claimed a seeding effect after segregating data by cloud-top temperature, estimated from 12-hourly rawinsondes. Their estimates did not take into account the high degree of variability of cloud-top temperature and their results should also be considered questionable.

The Bridger Range experiment exploratory analyses first used 24-h experimental units (Super and Heimbach 1983). In later analyses, these were subdivided into 6-h increments (Super 1986; Super and Heimbach 2009). A post hoc analysis using gauge data collected upwind of the seeding and crosswind (outside the target area) as a control yielded double-ratio estimates of 15% more precipitation due to seeding in the target area, provided temperatures were less than -9° C at levels about In more recent times, a concerted effort has been undertaken to improve experimental design and statistical analysis to study natural and seeded cloud systems and to verify basic components of the hypothesis (e.g., Gabriel 1999; Manton et al. 2011; Breed et al. 2014). The primary method to evaluate the effect of cloud seeding on seasonal enhancement of the snowpack has been with precipitation gauge measurements using target/control statistics.

The exploratory Snowy Mountains experiment in Australia (1955–63, Ryan and King 1997) used two areas, one target and one control, and during any 10-15 day period a random process determined whether clouds over the target area should or should not be seeded. In some experiments, a crossover design was employed where a random process was used to select which area would be the target and which would be the control for each period. The results produced statistically significant evidence for precipitation increases over the entire experiment, with a 19% increase significant at the 5% level. In the confirmatory Snowy Mountains experiment (2005-09, Manton et al. 2011; Manton and Warren 2011), 107 experimental units were obtained, yielding a positive, but not statistically significant, precipitation impact. A post hoc analysis identified that a source of uncertainty was introduced by units where the seeding generators operated for relatively few hours. Reanalysis with only units where generators collectively burned for more than 45 h showed an increase in primary target area precipitation of 14% at the 8% significance level and in the larger overall target area of 14% at the 3% significance level. Restratification of data introduced a potential error because of multiplicity, that is, finding a positive result by accident and not design. To avoid multiplicity, a third experiment (Manton et al. 2017), was carried out between 2010 and 2013 that had a dual-target area, wherein criteria for seeding were further modified a priori based on additional analysis and subsequent restratification of the Manton et al. (2011) and Manton and Warren (2011) datasets. The start criteria for an experimental unit in the 2010-13 experiment were restricted to minimize the chance of a unit having low precipitation or low wind speed, situations that produced little seeding effect in the previous experiment. The calculated increase in precipitation in this confirmatory experiment under the revised criteria ranged from 12% to 16% in the two areas, significant at the 6% level in the south, and 3% level in the north target areas. The study concluded that there was strong evidence of a positive seeding impact across the overall target area of the Snowy Mountains.

The Wyoming Weather Modification Pilot Program (WWMPP) randomized confirmatory seeding experiment was conducted between 2008 and 2014 over two mountain ranges in southern Wyoming. This experiment was preceded by an exploratory experiment conducted by Weather Modification, Inc., from 2004 to 2005 (Weather Modification, Inc. 2005) that indicated there was potential for precipitation enhancement by orographic cloud seeding. To validate this result, a confirmatory experiment was designed and initiated in 2008 (Breed et al. 2014). The confirmatory WWMPP experiment collected 118 randomized statistical experiment (RSE) precipitation cases that met qualitycontrol and seeding criteria over six years. The RSE included a crossover design between the two mountain ranges, whereas one served as the target and the other as the control, and the range to be targeted with seeding was chosen randomly (Breed et al. 2014). Each mountain range had a single target gauge site that served as a control site when the mountain barrier was not seeded, as well as two covariate sites. All gauge sites included three precipitation gauges to aid in the quality control process. The target gauges were located in forest clearings in which measured wind speeds were typically less than $2 \,\mathrm{m \, s}^{-1}$. Gauge undercatch due to wind becomes less than 15% at these wind speeds based on recent result from the WMO Solid Precipitation Intercomparison Project (SPICE) and other studies (Rasmussen et al. 2012), and thus no wind corrections were made.

The statistical results were inconclusive in that there was insufficient statistical evidence (p value of 0.28) to reject the null hypothesis that there was no effect of ground seeding after 118 randomized cases. The researchers concluded that it may be possible in their experiment to statistically reject the null hypothesis by collecting on the order of 1000 cases for a p value of 0.05 (i.e., a 5% chance that the precipitation change was not due to seeding), but that the time and expense required to collect such measurements made this prohibitive (Rasmussen et al. 2018).

Finally, statistical analyses of *operational* programs' impact on orographic snowpack enhancement appear both in nonreviewed reports and peer-reviewed articles. Authors have either been independent investigators (e.g., Silverman 2007, 2008, 2009), utilities sponsoring operations, or contractors doing the seeding (e.g., Henderson 1966, 2003a,b; Griffith and Solak 2002). Because of potential conflict of interest concerns, contractor assessments and utility self-assessments are not considered in this review. We focus only on independent evaluations.

2128

The approach to evaluating the effectiveness of operational cloud seeding, where programs typically do not employ randomized seeding techniques, has been to use a target-control approach where one basin is seeded throughout a winter season, while a neighboring basin remains unseeded as the control. Assessments have been made using a variety of metrics from precipitation gauge data, snow-water equivalent measurements, or streamflow measurements. The foundation for this type of assessment is that the metric of choice in the target and control basins is highly correlated so that the control basin behavior can predict what the natural behavior would have been in the target basin. Silverman (2007, 2008, 2009), for example, conducted a posteriori evaluations of nonrandomized operational seeding programs conducted in the Kern River, Kings River, and San Joaquin River basins of California using this approach. Semiannual and annual integrated streamflow measurements in the seeded basin catchment were used, with a neighboring unseeded basin catchment as a control. He noted that streamflow measurements have the advantage of long records and high correlation across basins in unseeded years. For the Kern River, ratio statistic evaluation of water year streamflow for a specific streamflow gauge chosen a priori for evaluation indicated a + 12.2% increase in streamflow due to seeding with 90% confidence that the true effect of seeding was between +6.1% and +18.6%. The probability that the effect of seeding is equal to or greater than 0% and 1% (the estimated threshold of costeffective operations) was 99.9. Similar results were obtained for the Kings River program, while weaker results suggesting an increase in precipitation were obtained in the San Joaquin operational program watershed. Silverman cautioned that his results should be taken as measures of the strength of the suggested seeding effect and not as measures of statistical significance. Simpler statistical approaches, for example applying traditional historical regression methods of comparing seasonal precipitation in seeded years with a regression line developed from data from target and control basins in nonseeded years, have also been used for operational programs. Such data must be viewed with caution, since the relationship between the target and control areas may have changed over time, or other meteorological factors, such as early snowmelt at selected measurement sites, may influence results (e.g., Super and Heimbach 2003).

4. Modeling evaluations of the underlying hypothesis of orographic cloud seeding

Evaluating seeding effects on orographic precipitation purely by observations and statistical analyses has always been challenging due to the unavailability of a control in nature, the difficulty in detecting the seeding signal and seeding effects on cloud and precipitation development, and the cost of conducting long term randomized seeding experiments. Thus, numerical models capable of simulating natural orographic cloud processes and cloud-seeding physics, and validated by observations, have become a useful tool to assess and quantify cloudseeding effects.

a. Natural precipitation

Modeling studies of natural orographic clouds have developed in concert with physical process studies. Investigations of natural clouds have focused on aerosol/cloud microphysical controls on the distribution of precipitation across mountain ranges (Cotton et al. 1986; Colle and Zeng 2004a,b; Colle et al. 2005a,b; Lin and Colle 2009; Saleeby et al. 2009, 2011, 2013), terraininduced local dynamics (Smith and Barstad 2004; Garvert et al. 2005a; Colle et al. 2008; Ikeda et al. 2010; Minder et al. 2008), and dynamical controls on precipitation estimation (Colle and Mass 2000; Garvert et al. 2005b). High-resolution studies of orographic precipitation (Ikeda et al. 2010; Rasmussen et al. 2011; Liu et al. 2017; Jing et al. 2017) have shown that cold-season natural precipitation in complex terrain can now be reasonably simulated with modern mesoscale models using the most advanced microphysical schemes, resulting in a seasonal precipitation dataset that can be superior to gridded gauge-based datasets for hydrological and other studies (e.g., Wrzesien et al. 2019). These studies show that natural seasonal snowfall over the mountain ranges of the western United States can be simulated to within 10% of SNOTEL observations if the model horizontal resolution is less than 6 km (Fig. 6). The success of these studies helped inspired the application of high-resolution model ensembles in the evaluation of recent orographic seeding programs (Rasmussen et al. 2018).

b. Seeded precipitation

Many numerical investigations on glaciogenic seeding effects in different cloud types have been performed (e.g., Reisin et al. 1996; Yin et al. 2000; Guo et al. 2006; Curic et al. 2007; Chen and Xiao 2010) using different AgI nucleation parameterizations and/or different model setups. Seeding effects on precipitation enhancement were found to be positive in most studies and seeding was shown to change the precipitation amount/distribution most of the time. However, most of these studies focused on deep-convective cloud-seeding scenarios. Few studies examined the sensitivities of seeding effects on wintertime orographic clouds to meteorological conditions and cloud microphysical properties in a



FIG. 6. Monthly time series of accumulative precipitation (mm) for (a) 2001–02, (b) 2003–04, (c) 2005–06, and (d) 2007–08. Solid line is simulated precipitation with the WRF Model at SNOTEL mountain site locations. Dashed lines are SNOTEL measurements, with gray shades representing 1 standard deviation from the average daily precipitation totals at SNOTEL sites. The dots are Parameter–Elevation Regressions on Independent Slopes Model (PRISM) monthly averaged snowfall estimates. [From Rasmussen et al. (2011).]

systematic way, Li and Pitter (1997) being the most thorough sensitivity study.

The laboratory observations of **DeMott** (1995) are the most complete data describing AgI accumulated nucleation rates as a function of temperature and saturation ratios with respect to ice and water for all four modes of nucleation (deposition, condensation freezing, immersion freezing, and contact freezing). Meyers et al. (1995) implemented an AgI nucleation parameterization based on DeMott (1995) into the Regional Atmospheric Modeling System and simulated an orographic cloud-seeding event from the Sierra Cooperative Pilot Project (Reynolds and Dennis 1986). Their results suggested that the model could simulate the microphysical interactions between AgI and cloud hydrometeors. Meyers et al. (1995) were the first to employ a 3D simulation of seeding effects in orographic clouds. Their study reproduced the primary results of the Sierra case study reported by Deshler et al. (1990). Meyers et al. (1995) also reviewed results of earlier 2D models examining seeding effects reported by

Hobbs et al. (1973), Young (1974), Plooster and Fukuta (1975), and Blumenstein et al. (1987).

More recently, Xue et al. (2013a) developed a microphysical parameterization that simulates the physical processes associated with AgI seeding. This parameterization was initially built upon the Thompson et al. (2008) bulk microphysics scheme in the Weather Research and Forecasting (WRF) Model and includes (i) AgI particle dispersion over complex terrain, (ii) the coagulation of AgI particles after they are emitted from ground-based or airborne generators, (iii) wet deposition of AgI through scavenging, (iv) dry deposition of AgI through fallout and canopy interception, (v) water activation of AgI particles associated with any soluble portion of the aerosol, (vi) ice nucleation of AgI particles through different modes, and (vii) cloud processing of AgI and regeneration of AgI particles upon evaporation and sublimation of hydrometeors (Fig. 7). The critical physical processes that determine seeding impact to the first order are dispersion of AgI in complex



FIG. 7. Schematic of the AgI-cloud interactions that are simulated in the seeding parameterization. [Adapted from Xue et al. (2013a), with additional parameterizations now included in the scheme.]

terrain and cloud-seeding microphysics (processes in Fig. 7). Simulating all of these processes remains challenging both scientifically and computationally even with the most advanced, state-of-the-art numerical modeling frameworks.

Nonetheless, the examination of the chain of events associated with glaciogenic seeding of orographic clouds has recently become possible due to the emergence of such state-of-the-art cloud-seeding microphysics parameterizations incorporating these processes (Xue et al. 2013a,b; Geresdi et al. 2017), advances in microphysical schemes for numerical models (Thompson et al. 2008; Thompson and Eidhammer 2014; Morrison and Grabowski 2008; Geresdi et al. 2014; Sarkadi et al. 2016), and sufficient computing power to resolve large eddies. The Xue et al. (2013a) AgI cloud-seeding parameterization, which was built on a bulk cloud microphysics scheme, tracks the conserved AgI number and mass within different hydrometeors to determine AgI-produced precipitation. This bulk microphysics cloud-seeding modeling framework has been used to investigate the microphysical chain of events of glaciogenic seeding and its effect on wintertime orographic clouds under

both idealized and realistic conditions (Xue et al. 2013a,b; Chu et al. 2014; Xue et al. 2016; Chu et al. 2017a,b; Xue et al. 2017). The scheme still requires validation, a primary driver for the recent SNOWIE field program (Tessendorf et al. 2019).

c. Wyoming case study of orographic cloud seeding with observations and model

During ASCII, a ground-based cloud-seeding event occurred on 18 February 2009 in the Medicine Bow Mountains of southern Wyoming. The event was thoroughly studied using observed soundings, liquid water path from radiometers, and vertical Doppler velocity and reflectivity-based cloud structures observed by the WCR. In addition, LES (100-m grid spacing) and non-LES simulations (900-m grid spacing) of this case were performed with and without the AgI cloud-seeding parameterization (Xue et al. 2013a,b, 2016). The 100-m LES was able to reproduce the thermodynamics and flow pattern, including boundary layer turbulence and a hydraulic jump in the lee. It also captured the spatial distribution of SLW and the cloud vertical structure, but had some difficulty simulating the cloud evolution from the untreated to the treated period (Chu et al. 2014; Xue et al. 2016). Model-simulated cloud-seeding impacts on clouds were found to occur primarily within the PBL (below 1 km AGL).

d. Idaho case study of orographic cloud seeding with observation and model

Four cloud-seeding cases in the Payette river basin of western Idaho were simulated from a snow trace chemistry field campaign that took place between 2003 and 2005 (Xue et al. 2017). Two ground-based and two airborne events were simulated by the WRF Model in LES mode with a grid spacing of 667 m. Testing indicated that grid spacings finer than 900 m were sufficient to resolve the most important turbulent scale that shapes the PBL structure. For all simulations performed for orographic clouds over complex terrain, the vertical coordinate had a grid spacing smaller than 100 m in the lowest 3 km AGL.

The 666-m LES results were compared with the observed soundings, precipitation from SNOTEL sites, and silver mass concentrations within the snow measured from snow samples by plasma mass spectrometers. The comparisons showed that the silver mass concentration accumulated in simulated snowfall had the same order of magnitude as the observations from all sample sites. Furthermore, the predicted spatial distributions of the silver mass concentrations agreed qualitatively well with the observations.

e. Ensemble modeling approach

LES is a useful modeling tool for the seeding simulation that resolves the AgI dispersion in complex terrain. However, the extremely high computational cost associated with LES prevents it from being used for large domains, long periods, or multiple cases simulations. As an alternative, non-LES simulations running PBL schemes have to be used in these situations. For example, to help understand and interpret the statistical results of the WWMPP (Rasmussen et al. 2018), numerical modeling simulations of the impact of cloud seeding were conducted for each of the 118 RSE cases.

Given the uncertainties associated with numerical modeling, an ensemble cloud-seeding simulation approach was designed to evaluate the WWMPP program using the WRF Model with the cloud-seeding parameterization (Rasmussen et al. 2018). The ensemble design was based upon varying numerous potential sources of uncertainty in simulating orographic clouds, precipitation, and associated seeding effects. These sources of uncertainty include: large-scale driving meteorological conditions, model physics for both natural clouds and seeding processes, background CCN and ice nuclei (IN)

concentrations, and spatial/temporal distributions of precipitation. A 72-member ensemble was designed for each of the 118 RSE cases during the WWMPP (Rasmussen et al. 2018). The ensemble mean precipitation validated well against observations. The simulated seeding effects, defined as the difference of precipitation from simulations with and without seeding, indicated a median 5% enhancement of precipitation relative to seedable storms (those containing SLW) over a given winter season, with an inner quartile range of 3%–7% for the Sierra Madre and Medicine Bow Ranges in Wyoming based on about 9000 seeding case simulations. If one considers all storms during the six winter seasons, the median seeding effect was 1.5% increase in precipitation over that occurring naturally.

5. Recent advances in assessment and operations

a. Advances in modeling

A major advance in snowfall simulation in orographic environments has been the demonstration that seasonal natural orographic precipitation can now be realistically simulated within 10% of observations (Ikeda et al. 2010; Rasmussen et al. 2011; Liu et al. 2017). This presents potential opportunities for modeling the impacts of season-long operational or research-based orographic cloud seeding. Prior to this capability, uncertainty in model estimates of natural mountain precipitation largely prevented scientists from applying state of the art models to orographic cloud seeding. Although the sophistication of modern mesoscale models and cloud physics parameterizations (Rasmussen et al. 2011) has allowed implementation of cloud seeding directly into a model, convincing tests of the ability of the model to accurately assess seeding impacts in seasonal seeding operations still are required in future experiments.

Another advancement that allows use of models to evaluate cloud seeding is the increase by two-three orders of magnitude in computer speed over the past 10 years. This now permits simulations at very high resolution and the application of ensemble approaches to evaluate the impact of cloud seeding, providing a range of uncertainty for seeding effects. The ensemble approach also allows evaluation of aspects of the model initialization and physics that are known to be uncertain, providing better estimates of the likely range of seeding impact (Rasmussen et al. 2018). These novel modeling approaches in weather modification research require extensive testing and validation with observations, a goal for future projects, before their results can be accepted as convincing evidence of orographic precipitation enhancement.

2132

b. Advances in observing technologies

The use of more sensitive, better resolved radars (such as airborne W-band radar or specially deployed X- and K-band radars) to investigate the physical effects of cloud seeding has made it possible to infer effects of ground-based seeding (Pokharel et al. 2017; Chu et al. 2017a) and directly detect plumes of ice particles created by airborne seeding (French et al. 2018). This, combined with specially deployed radiometers, rawinsondes, highresolution precipitation gauges, and in situ airborne measurements in seeded clouds has opened new opportunities for tracing the seeding process from the AgI source to arrival of the precipitation at the ground (Tessendorf et al. 2019). A deficiency in cloud-seeding evaluations remains ground validation of seeding effects in target experiments. Although season-long precipitation impacts of seeding have been estimated in many experiments, the impact on precipitation of the passage of a single plume of ice particles generated from an airborne seeding line or a ground generator remains difficult to assess. There is often not a clear seeding signature on radar (except in cases where the initial radar reflectivity is dominated by SLW, as in French et al. 2018). This is compounded by the measurement uncertainty of snow gauges and difficulty of making measurements in snowstorms in remote mountainous terrain. Yet it is precisely these kind of case-study measurements that are needed to validate numerical simulations of seeding impacts. One possibility currently under exploration involves hightemporal-resolution (e.g., 30 min) snow sample collection for chemistry analysis, but results are not available at the time of this writing. A key challenge remains extrapolating obvious local effects observed with radar to the demonstration of more widespread and longlasting impact on increasing snowfall across a target basin.

6. State of the science and a look to the future

Meteorological studies are continuing to advance our understanding of the distribution of supercooled water in orographic cloud systems, and the sources of vertical circulations that can lead to the production of that water. Unfortunately, a corresponding understanding of natural ice nucleation mechanisms in orographic clouds remains elusive. Future scientific studies in orographic environments should focus on this key issue. Our understanding of the growth and fallout of ice particles as precipitation over mountains has advanced considerably through detailed case studies. However, because cloudseeding effects represent at best a small percentage increase in precipitation, it is essential that future efforts focus on ground-based measurements of precipitation intensity, type, and chemistry, and ground-based remote sensing of SLW. These measurements have always been challenging because of the difficulty of working in complex terrain in winter.

Models have improved in sophistication to the point that AgI plume dispersion and the growth and fallout of precipitation produced by AgI aerosol can be simulated; however, verification studies are required to validate model results before the models can be used to quantitatively predict seeding outcomes. For example, data are required to validate model predictions related to time scale for nucleation and fallout of ice particles versus the time scale for the aerosol/ice particles to be transported out of a target basin. Despite all the uncertainties, models are now being used operationally to estimate likely seeding opportunities based on short-term forecasts. For example, Idaho Power's orographic cloudseeding operations are guided entirely by advanced WRF Model prediction of seeding windows, based on current understanding of seeding outcomes in various weather regimes.

Concerning seeding, clear physical evidence has been obtained that orographic clouds containing supercooled water, when seeded with AgI, produce plumes of ice particles that originate downwind of the seeding location and reach the ground through precipitation growth and fallout. The evidence, at present, comes from studies of clouds with sufficiently small concentrations of natural ice particles such that radar signals from ice particle plumes generated by seeding are not overwhelmed by background radar echoes from naturally occurring ice particle populations. Because sufficiently strong natural radar echoes are often produced by cloud systems that are deep and have cold cloud tops, clear physical evidence of seeding effects, at present, has been limited to shallow orographic clouds with warm $(>-15^{\circ}C)$ cloud-top temperatures; however, deep orographic clouds with strong radar echoes may still locally contain large amounts of SLW (section 2a).

A second independent line of evidence that cloud seeding has an impact on clouds has been the presence of trace silver concentrations in snow on the ground in the target area following seeding events. Experiments in which active ice-nucleating aerosol (AgI) and inert tracer aerosol (In_2O_3) were simultaneously released at collocated generators consistently demonstrate that trace silver in the snowpack following passage of a seeding event is due to ice nucleation rather than precipitation scavenging of aerosol. Experiments tying tracer detection in snow to model predictions of ice particle trajectories associated with seeding hold promise and should be further explored in future experiments.

Seven randomized statistical experiments have been conducted targeting orographic cloud systems. Unfortunately, four of them, all conducted in the 1960s-70s, had design flaws that rendered their findings questionable. Of the remaining three, two experiments, one in Australia's Snowy Mountains and the other in Wyoming, were confirmatory. The first, in Australia, reported statistically significant increases in precipitation. The second, in Wyoming's Medicine Bow Mountains, reported positive results but was unable to obtain statistical significance because of an insufficient number of events. Post hoc stratification analyses from the three valid experiments and from analyses of operational projects have consistently yielded positive, statistically significant precipitation increases. However, caution is necessary in interpreting these results, because the issue of multiplicity remains a concern when interpreting post hoc stratifications. Future statistical experiments should be closely coupled with observational studies employing the full suite of techniques now available for physical evaluations. Together, they hold promise for narrowing the uncertainty that has accompanied orographic cloudseeding research over its long history.

The need for water for urban use, agriculture, and power generation in the western United States and other arid regions of the world where water resources depend on melting of winter snowpack in high mountain ranges will increase in the future as the human population increases and Earth's changing climate alters the extent of these snowpack reservoirs. As a result, operational cloud seeding will continue unabated, despite uncertainty in the outcome of the seeding. Significant progress has been made over nearly seven decades in understanding the natural structure of orographic clouds systems and the impact of seeding. At the same time, technological advances such as computer modeling, computer speed, and networking and advanced airborne radar systems have allowed scientists to explore orographic cloud seeding in ways that were only dreamed of 20 years ago.

It is societally important that this research continue. To make progress going into the future, it is necessary to carry out a confirmatory statistical experiment backed by comprehensive meteorological observations and physical studies employing all of the technologies now available to unambiguously extract the seeding signatures. It is clear, however, that obtaining statistically significant results will require a project extending over a sufficiently long time period. Obtaining commitments to conduct extended projects remains a challenge. The most feasible approach for future studies would likely involve a well-designed experiment that couples and coordinates observational, statistical, and modeling research with a long-term operational project similar to the recent WWMPP/ASCII and SNOWIE experiments (Pokharel and Geerts 2016; Rasmussen et al. 2018; Tessendorf et al. 2019). Water needs in the future may provide the impetus for these types of projects to be realized.

Acknowledgments. The research underlying this review was supported under NSF Grants AGS-1547101, AGS-1546963, and AGS-1546939 and by the Idaho Power Company. The National Center for Atmospheric Research is sponsored by the National Science Foundation.

REFERENCES

- Aikins, J., K. Friedrich, B. Geerts, and B. Pokharel, 2016: Role of a low-level jet and turbulence on winter orographic snowfall. *Mon. Wea. Rev.*, **144**, 3277–3300, https://doi.org/10.1175/ MWR-D-16-0025.1.
- Barnes, H. C., J. P. Zagrodnik, L. A. McMurdie, A. K. Rowe, and R. A. Houze, 2018: Kelvin–Helmholtz waves in precipitating midlatitude cyclones. J. Atmos. Sci., 75, 2763–2785, https:// doi.org/10.1175/JAS-D-17-0365.1.
- Bergeron, T., 1935: On the physics of cloud and precipitation. Proc. Fifth Assembly U.G.G.I. Lisbon, Vol. 2, Lisbon, Portugal, Union Géodésique et Géophysique Internationale, 156–178.
- Blumenstein, R. R., R. M. Rauber, L. O. Grant, and W. G. Finnegan, 1987: Application of ice nucleation kinetics in orographic clouds. J. Climate Appl. Meteor., 26, 1363– 1376, https://doi.org/10.1175/1520-0450(1987)026<1363: AOINKI>2.0.CO;2.
- Boe, B. A., J. A. Heimbach Jr., T. W. Krauss, L. Xue, X. Chu, and J. T. McPartland, 2014: The dispersion of silver iodide particles from ground-based generators over complex terrain. Part I: Observations with acoustic ice nucleus counters. J. Appl. Meteor. Climatol., 53, 1325–1341, https://doi.org/10.1175/ JAMC-D-13-0240.1.
- Bradley, R. A., S. S. Srivastava, and A. Lanzdorf, 1979: Some approaches to statistical analysis of a weather modification experiment. *Commun. Stat. Theory Methods*, 8, 1049–1081, https://doi.org/10.1080/03610927908827815.
- Breed, D., R. Rasmussen, C. Weeks, B. Boe, and T. Deshler, 2014: Evaluating winter orographic cloud seeding: Design of the Wyoming Weather Modification Pilot Project (WWMPP). J. Appl. Meteor. Climatol., 53, 282–299, https://doi.org/ 10.1175/JAMC-D-13-0128.1.
- Bruintjes, R. T., 1999: A review of cloud seeding experiments to enhance precipitation and snow new prospects. *Bull. Amer. Meteor. Soc.*, **80**, 805–802, https://doi.org/10.1175/1520-0477(1999)080<0805:AROCSE>2.0.CO;2.
- —, T. L. Clark, and W. D. Hall, 1994: Interactions between topographic airflow and cloud/precipitation development during the passage of a winter storm in Arizona. J. Atmos. Sci., 51, 48–67, https://doi.org/10.1175/1520-0469(1994)051<0048: IBTAAC>2.0.CO:2.
- , —, and —, 1995: The dispersion of tracer plumes in mountainous regions in central Arizona: Comparisons between observations and modeling results. *J. Appl. Meteor.*, 34, 971–988, https://doi.org/10.1175/1520-0450(1995)034<0971: TDOTPI>2.0.CO;2.

JOURNAL OF APPLIED METEOROLOGY AND CLIMATOLOGY

- Chappell, C. F., L. O. Grant, and P. W. Mielke, 1971: Cloud seeding effects on precipitation intensity and duration of wintertime orographic clouds. J. Appl. Meteor., 10, 1006–1010, https:// doi.org/10.1175/1520-0450(1971)010<1006:CSEOPI>2.0.CO;2.
- Chen, B., and H. Xiao, 2010: Silver iodide seeding impact on the microphysics and dynamics of convective clouds in the high plains. *Atmos. Res.*, 96, 186–207, https://doi.org/10.1016/ j.atmosres.2009.04.001.
- Chow, F. K., A. P. Weigel, R. L. Street, M. W. Rotach, and M. Xue, 2006: High-resolution large-eddy simulations of flow in a steep Alpine valley. Part I: Methodology, verification, and sensitivity experiments. J. Appl. Meteor. Climatol., 45, 63–86, https://doi.org/10.1175/JAM2322.1.
- Chu, X., L. Xue, B. Geerts, R. Rasmussen, and D. Breed, 2014: A case study of radar observations and WRF LES simulations of the impact of ground-based glaciogenic seeding on orographic clouds and precipitation. Part I: Observations and model validations. J. Appl. Meteor. Climatol., 53, 2264–2286, https:// doi.org/10.1175/JAMC-D-14-0017.1.
- —, B. Geerts, L. Xue, and B. Pokharel, 2017a: A case study of cloud radar observations and large-eddy simulations of a shallow stratiform orographic cloud, and the impact of glaciogenic seeding. J. Appl. Meteor. Climatol., 56, 1285–1304, https://doi.org/10.1175/JAMC-D-16-0364.1.
- —, —, —, and R. Rasmussen, 2017b: Large-eddy simulations of the impact of ground-based glaciogenic seeding on shallow orographic convection: A case study. J. Appl. Meteor. Climatol., 56, 69–84, https://doi.org/10.1175/JAMC-D-16-0191.1.
- —, L. Xue, B. Geerts, and B. Kosovic, 2018: The impact of boundary layer turbulence on snow growth and precipitation: Idealized large eddy simulations. *Atmos. Res.*, **204**, 54–66, https://doi.org/10.1016/j.atmosres.2018.01.015.
- Colle, B. A., and C. F. Mass, 2000: The 5–9 February 1996 flooding event over the Pacific Northwest: Sensitivity studies and evaluation of the MM5 precipitation forecasts. *Mon. Wea. Rev.*, **128**, 593–617, https://doi.org/10.1175/1520-0493(2000) 128<0593:TFFEOT>2.0.CO;2.
- —, and Y. Zeng, 2004a: Bulk microphysical sensitivities within the MM5 for orographic precipitation. Part I: The Sierra 1986 event. *Mon. Wea. Rev.*, **132**, 2780–2801, https://doi.org/ 10.1175/MWR2821.1.
- —, and —, 2004b: Bulk microphysical sensitivities within the MM5 for orographic precipitation. Part II: Impact of barrier width and freezing level. *Mon. Wea. Rev.*, **132**, 2802–2815, https://doi.org/10.1175/MWR2822.1.
- —, M. F. Garvert, J. B. Wolfe, C. F. Mass, and C. P. Woods, 2005a: The 13–14 December 2001 IMPROVE-2 event. Part III: Simulated microphysical budgets and sensitivity studies. *J. Atmos. Sci.*, **62**, 3535–3558, https://doi.org/10.1175/ JAS3552.1.
- —, J. B. Wolfe, W. J. Steenburgh, D. E. Kingsmill, J. A. W. Cox, and J. C. Shafer, 2005b: High-resolution simulations and microphysical validation of an orographic precipitation event over the Wasatch Mountains during IPEX IOP3. *Mon. Wea. Rev.*, 133, 2947–2971, https://doi.org/10.1175/MWR3017.1.
- —, Y. Lin, S. Medina, and B. F. Smull, 2008: Orographic modification of convection and flow kinematics by the Oregon Coast Range and Cascades during IMPROVE-2. *Mon. Wea. Rev.*, **136**, 3894–3916, https://doi.org/10.1175/ 2008MWR2369.1.
- Conrick, R., C. F. Mass, and Q. Zhong, 2018: Simulated Kelvin-Helmholtz Waves over terrain and their microphysical

implications. J. Atmos. Sci., **75**, 2787–2800, https://doi.org/ 10.1175/JAS-D-18-0073.1.

- Cooper, W. A., and C. P. R. Saunders, 1980: Winter storms over the San Juan Mountains. Part II: Microphysical processes. J. Appl. Meteor., **19**, 927–941, https://doi.org/10.1175/1520-0450(1980) 019<0927:WSOTSJ>2.0.CO;2.
- —, and G. Vali, 1981: The origin of ice in mountain cap clouds. J. Atmos. Sci., 38, 1244–1259, https://doi.org/10.1175/1520-0469(1981)038<1244:TOOIIM>2.0.CO;2.
- Cotton, W. R., G. J. Tripoli, R. M. Rauber, and E. A. Mulvihill, 1986: Numerical simulation of the effects of varying ice crystal nucleation rates and aggregation processes on orographic snowfall. J. Climate Appl. Meteor., 25, 1658–1680, https:// doi.org/10.1175/1520-0450(1986)025<1658:NSOTEO>2.0.CO;2.
- Curic, M., D. Janc, and V. Vuckovic, 2007: Cloud seeding impact on precipitation as revealed by cloud-resolving mesoscale model. *Meteor. Atmos. Phys.*, 95, 179–193, https://doi.org/10.1007/ s00703-006-0202-y.
- DeMott, P. J., 1995: Quantitative descriptions of ice formation mechanisms of silver iodide-type aerosols. *Atmos. Res.*, 38, 63– 99, https://doi.org/10.1016/0169-8095(94)00088-U.
- Demoz, B. B., R. Zhang, and R. L. Pitter, 1993: An analysis of Sierra Nevada winter orographic storms: Ground-based ice-crystal observations. J. Appl. Meteor., 32, 1826–1836, https://doi.org/ 10.1175/1520-0450(1993)032<1826:AAOSNW>2.0.CO;2.
- Deshler, T., and D. W. Reynolds, 1990: The persistence of seeding effects in a winter orographic cloud seeded with silver iodide burned in acetone. J. Appl. Meteor., 29, 477–488, https:// doi.org/10.1175/1520-0450(1990)029<0477:TPOSEI>2.0.CO;2.
- —, —, and A. W. Huggins, 1990: Physical response of winter orographic clouds over the Sierra Nevada to airborne seeding using dry ice or silver iodide. *J. Appl. Meteor.*, **29**, 288–330, https:// doi.org/10.1175/1520-0450(1990)029<0288:PROWOC>2.0.CO;2.
- Elliott, R. D., 1986: Review of wintertime orographic cloud seeding. Precipitation Enhancement—A Scientific Challenge, Meteor. Monogr., No. 43, Amer. Meteor. Soc., 87–104, https:// doi.org/10.1175/0065-9401-21.43.87.
- —, R. W. Shaffer, A. Court, and J. F. Hannaford, 1978: Randomized cloud seeding in the San Juan Mountains, Colorado. *J. Appl. Meteor.*, **17**, 1298–1318, https://doi.org/10.1175/1520-0450(1978)017<1298:RCSITS>2.0.CO;2.
- Findeisen, W., 1938: Die kolloidmeteorologischen Vorgänge bei der Niederschlagsbildung (Colloidal meteorological processes in the formation of precipitation). *Meteor. Z.*, 55, 121–133, https://doi.org/10.1127/metz/2015/0675.
- Fisher, J. M., M. L. Lytle, M. L. Kunkel, D. Blestrud, V. P. Holbrook, S. K. Parkinson, P. R. Edwards, and S. G. Benner, 2016: Evaluation of glaciogenic cloud seeding using trace chemistry. J. Wea. Modif., 48, 24–42.
- —, —, —, D. R. Blestrud, N. W. Dawson, S. K. Parkinson, R. Edwards, and S. G. Benner, 2018: Assessment of groundbased and aerial cloud seeding using trace chemistry. *Adv. Meteor.*, 2018, 7293987, https://doi.org/10.1155/2018/7293987.
- French, J. R., and Coauthors, 2018: Precipitation formation from orographic cloud seeding. *Proc. Natl. Acad. Sci. USA*, **115**, 1168–1173, https://doi.org/10.1073/pnas.1716995115.
- Gabriel, K. R., 1995: Climax again? J. Appl. Meteor., 34, 1225–1227, https://doi.org/10.1175/1520-0450(1995)034<1225:CA>2.0.CO;2.
- —, 1999: Ratio statistics for randomized experiments in precipitation stimulation. J. Appl. Meteor., 38, 290–301, https:// doi.org/10.1175/1520-0450(1999)038<0290:RSFREI>2.0.CO;2.
- Gagin, A., and J. Neumann, 1981: The Second Israeli Randomized Cloud Seeding Experiment: Evaluation of the results.

J. Appl. Meteor., **20**, 1301–1311, https://doi.org/10.1175/1520-0450(1981)020<1301:TSIRCS>2.0.CO;2.

- Garstang, M., R. Bruintjes, R. Serafin, H. Orville, B. Boe, W. Cotton, and J. Warburton, 2005: Finding common ground. *Bull. Amer. Meteor. Soc.*, 86, 647–655, https://doi.org/10.1175/ BAMS-86-5-647.
- Garvert, M. F., B. A. Colle, and C. F. Mass, 2005a: The 13–14 December 2001 IMPROVE-2 event. Part I: Synoptic and mesoscale evolution and comparison with a mesoscale model simulation. J. Atmos. Sci., 62, 3474–3492, https://doi.org/10.1175/ JAS3549.1.
- —, C. P. Woods, B. A. Colle, C. F. Mass, P. V. Hobbs, M. T. Stoelinga, and J. B. Wolfe, 2005b: The 13–14 December 2001 IMPROVE-2 event. Part II: Comparisons of MM5 model simulations of clouds and precipitation with observations. *J. Atmos. Sci.*, **62**, 3520–3534, https://doi.org/10.1175/JAS3551.1.
- Geerts, B., and Q. Miao, 2010: Vertically pointing airborne Doppler radar observations of Kelvin–Helmholtz billows. *Mon. Wea. Rev.*, **138**, 982–986, https://doi.org/10.1175/ 2009MWR3212.1.
- —, —, Y. Yang, R. Rasmussen, and D. Breed, 2010: An airborne profiling radar study of the impact of glaciogenic cloud seeding on snowfall from winter orographic clouds. J. Atmos. Sci., 67, 3286–3302, https://doi.org/10.1175/2010JAS3496.1.
- —, —, and —, 2011: Boundary layer turbulence and orographic precipitation growth in cold clouds: Evidence from profiling airborne radar data. J. Atmos. Sci., 68, 2344–2365, https://doi.org/10.1175/JAS-D-10-05009.1.
- —, and Coauthors, 2013: The AgI Seeding Cloud Impact Investigation (ASCII) campaign 2012: Overview and preliminary results. J. Wea. Modif., 45, 24–43.
- —, Y. Yang, R. Rasmussen, S. Haimov, and B. Pokharel, 2015a: Snow growth and transport patterns in orographic storms as estimated from airborne vertical-plane dual-Doppler radar data. *Mon. Wea. Rev.*, 143, 644–665, https://doi.org/10.1175/MWR-D-14-00199.1.
- —, B. Pokharel, and D. Kristovich, 2015b: Blowing snow as a natural glaciogenic cloud seeding mechanism. *Mon. Wea. Rev.*, 143, 5017–5033, https://doi.org/10.1175/MWR-D-15-0241.1.
- Geresdi, I., N. Sarkadi, and G. Thompson, 2014: Effect of the accretion by water drops on the melting of snowflakes. *Atmos. Res.*, 149, 96–110, https://doi.org/10.1016/j.atmosres.2014.06.001.
- —, L. Xue, and R. Rasmussen, 2017: Evaluation of orographic cloud seeding using a bin microphysics scheme: Two-dimensional approach. J. Appl. Meteor. Climatology, 56, 1443–1462, https:// doi.org/10.1175/JAMC-D-16-0045.1.
- Grant, L. O., and R. E. Elliott, 1974: The cloud seeding temperature window. J. Appl. Meteor., 13, 355–363, https://doi.org/ 10.1175/1520-0450(1974)013<0355:TCSTW>2.0.CO;2.
- Griffith, D. A., and M. E. Solak, 2002: Economic feasibility assessment of winter cloud seeding in the Boise River drainage, Idaho. J. Wea. Modif., 34, 39–46.
- Gultepe, I., 2015: Mountain weather: Observations and modeling. Advances in Geophysics, Vol. 56, Academic Press, 229–312, https://doi.org/10.1016/bs.agph.2015.01.001.
- —, G. A. Isaac, P. Joe, P. Kucera, J. Thériault, and T. Fisico, 2014: Roundhouse (RND) mountain top research site: Measurements and uncertainties for winter alpine weather conditions. *Pure Appl. Geophys.*, **171**, 59–85, https://doi.org/ 10.1007/s00024-012-0582-5.
- —, R. Rabin, R. Ware, and M. Pavolonis, 2016: Light snow precipitation and effects on weather and climate. *Advances in Geophysics*, Vol. 57, Academic Press, 147–210, https://doi.org/ 10.1016/bs.agph.2016.09.001.

- Guo, X., G. Zheng, and D. Jin, 2006: A numerical comparison study of cloud seeding by silver iodide and liquid carbon dioxide. *Atmos. Res.*, **79**, 183–226, https://doi.org/10.1016/j.atmosres.2005.04.005.
- Hallett, J., and S. C. Mossop, 1974: Production of secondary ice particles during the riming process. *Nature*, 249, 26–28, https:// doi.org/10.1038/249026a0.
- Haupt, S. E., R. M. Rauber, B. Carmichael, J. C. Knievel, and J. L. Cogan, 2019: 100 years of progress in applied meteorology. Part I: Basic applications. A Century of Progress in Atmospheric and Related Sciences: Celebrating the American Meteorological Society Centennial, Meteor. Monogr., No. 59, https://doi.org/10.1175/AMSMONOGRAPHS-D-18-0004.1.
- Heggli, M. F., and R. M. Rauber, 1988: The characteristics and evolution of supercooled water in wintertime storms over the Sierra Nevada: A summary of microwave radiometric measurements taken during the Sierra Cooperative Pilot Project. J. Appl. Meteor., 27, 989–1015, https://doi.org/10.1175/1520-0450(1988)027<0989:TCAEOS>2.0.CO;2.
- —, L. Vardiman, R. E. Stewart, and A. Huggins, 1983: Supercooled liquid water and ice crystal distributions within Sierra Nevada winter storms. J. Climate Appl. Meteor., 22, 1875–1886, https://doi.org/10.1175/1520-0450(1983)022<1875: SLWAIC>2.0.CO;2.
- Henderson, T. J., 1966: A ten year non-randomized cloud seeding program on the Kings River in California. J. Appl. Meteor., 5, 697–702, https://doi.org/10.1175/1520-0450(1966)005<0697: ATYNRC>2.0.CO;2.
- —, 2003a: The Kings River Weather Resources Management Program. J. Wea. Modif., 35, 45–51.
- —, 2003b: New assessment of the economic impacts from six winter snowpack augmentation projects. J. Wea. Modif., 35, 41–44.
- Hill, G. E., 1980: Seeding-opportunity recognition in winter orographic clouds. J. Appl. Meteor., 19, 1371–1381, https://doi.org/ 10.1175/1520-0450(1980)019<1371:SORIWO>2.0.CO;2.
- —, and D. S. Woffinden, 1980: A balloonborne instrument for the measurement of vertical profiles of supercooled liquid water concentration. J. Appl. Meteor., 19, 1285–1292, https://doi.org/ 10.1175/1520-0450(1980)019<1285:ABIFTM>2.0.CO;2.
- Hobbs, P. V., 1975a: The nature of winter clouds and precipitation in the Cascade Mountains and their modification by artificial seeding. Part I: Natural conditions. J. Appl. Meteor., 14, 783–804, https://doi.org/10.1175/1520-0450(1975)014<0783: TNOWCA>2.0.CO;2.
- —, 1975b: The nature of winter clouds and precipitation in the Cascade Mountains and their modification by artificial seeding. Part III: Case studies of the effects of seeding. J. Appl. Meteor., 14, 819–858, https://doi.org/10.1175/1520-0450(1975) 014<0819:TNOWCA>2.0.CO;2.
- —, and A. L. Rangno, 1979: Comments on the Climax and Wolf Creek Pass cloud seeding experiments. J. Appl. Meteor., 18, 1233–1237, https://doi.org/10.1175/1520-0450(1979)018<1233: COTCAW>2.0.CO;2.
- —, R. C. Easter, and A. B. Fraser, 1973: A theoretical study of the flow of air and fallout of solid precipitation over mountainous terrain: Part II. Microphysics. J. Atmos. Sci., 30, 813–823, https:// doi.org/10.1175/1520-0469(1973)030<0813:ATSOTF>2.0.CO;2.
- —, J. H. Lyons, J. D. Locatelli, K. R. Biswas, L. F. Radke, R. R. Weiss Sr., and A. L. Rangno, 1981: Radar detection of cloud-seeding effects. *Science*, **213**, 1250–1252, https://doi.org/ 10.1126/science.213.4513.1250.
- Holroyd, E. W., J. T. McPartland, and A. B. Super, 1988: Observations of silver iodide plumes over the Grand Mesa of

Colorado. J. Appl. Meteor., **27**, 1125–1144, https://doi.org/ 10.1175/1520-0450(1988)027<1125:OOSIPO>2.0.CO;2.

- Houze, R. A., Jr., and S. Medina, 2005: Turbulence as a mechanism for orographic precipitation enhancement. J. Atmos. Sci., 62, 3599–3623, https://doi.org/10.1175/JAS3555.1.
- Huggins, A. W., 2007: Another wintertime cloud seeding case study with strong evidence of seeding effects. J. Wea. Modif., 39, 9–36.
- —, 2008: Summary of studies that document the effectiveness of cloud seeding for snowfall augmentation. J. Wea. Modif., 40, 119–126.
- Ikeda, K., R. M. Rasmussen, W. D. Hall, and G. Thompson, 2007: Observations of freezing drizzle in extratropical cyclonic storms during IMPROVE-2. J. Atmos. Sci., 64, 3016–3043, https://doi.org/10.1175/JAS3999.1.
- —, and Coauthors, 2010: Simulation of seasonal snowfall over Colorado. *Atmos. Res.*, 97, 462–477, https://doi.org/10.1016/ j.atmosres.2010.04.010.
- Jing, X., and B. Geerts, 2015: Dual-polarization radar data analysis of the impact of ground-based glaciogenic seeding on winter orographic clouds. Part II: Convective clouds. J. Appl. Meteor. Climatol., 54, 2099–2117, https://doi.org/10.1175/JAMC-D-15-0056.1.
 - —, —, K. Friedrich, and B. Pokharel, 2015: Dual-polarization radar data analysis of the impact of ground-based glaciogenic seeding on winter orographic clouds. Part I: Mostly stratiform clouds. J. Appl. Meteor. Climatol., 54, 1944–1969, https:// doi.org/10.1175/JAMC-D-14-0257.1.
- —, —, Y. Wang, and C. Liu, 2017: Evaluating seasonal orographic precipitation in the interior western United States using gauge data, gridded precipitation estimates, and a regional climate simulation. J. Hydrometeor., 18, 2541–2558, https://doi.org/10.1175/JHM-D-17-0056.1.
- Keeler, J. M., B. F. Jewett, R. M. Rauber, G. M. McFarquhar, R. M. Rasmussen, L. Xue, C. Liu, and G. Thompson, 2016a: Dynamics of cloud-top generating cells in winter cyclones. Part I: Idealized simulations in the context of field observations. *J. Atmos. Sci.*, **73**, 1507–1527, https://doi.org/10.1175/JAS-D-15-0126.1.

-, —, —, —, , —, , —, , and —, 2016b: Dynamics of cloud-top generating cells in winter cyclones. Part II: Radiative and instability forcing. *J. Atmos. Sci.*, **73**, 1529–1553, https://doi.org/10.1175/JAS-D-15-0127.1.

- Kenny, J. F., N. L. Barber, S. S. Hutson, K. S. Linsey, J. K. Lovelace, and M. A. Maupin, 2009: Estimated use of water in the United States in 2005. U.S. Geological Survey Circular 1344, 52 pp., https://pubs.usgs.gov/circ/1344/pdf/c1344.pdf.
- Kochendorfer, J., and Coauthors, 2017: Errors and adjustments for single-Alter-shielded and unshielded weighing gauge precipitation measurements from WMO-SPICE. *Hydrol. Earth Syst. Sci.*, **21**, 3525–3542, https://doi.org/10.5194/hess-21-3525-2017.
- Kumjian, M. R., S. A. Rutledge, R. M. Rasmussen, P. C. Kennedy, and M. Dixon, 2014: High-resolution polarimetric radar observations of snow-generating cells. *J. Appl. Meteor. Climatol.*, 53, 1636–1658, https://doi.org/10.1175/JAMC-D-13-0312.1.
- Kusunoki, K., M. Murakami, M. Hoshimoto, N. Orikasa, Y. Yamada, H. Mizuno, K. Hamazu, and H. Watanabe, 2004: The characteristics and evolution of orographic snow clouds under weak cold

advection. *Mon. Wea. Rev.*, **132**, 174–191, https://doi.org/10.1175/ 1520-0493(2004)132<0174:TCAEOO>2.0.CO;2.

- —, and Coauthors, 2005: Observations of quasi-stationary and shallow orographic snow clouds: Spatial distributions of supercooled liquid water and snow particles. *Mon. Wea. Rev.*, **133**, 743–751, https://doi.org/10.1175/MWR2874.1.
- Lee, T. F., 1988: Winter diurnal trends of Sierra Nevada supercooled liquid water and precipitation. J. Appl. Meteor., 27, 458–472, https://doi.org/10.1175/1520-0450(1988)027<0458: WDTOSN>2.0.CO;2.
- Li, Z., and R. L. Pitter, 1997: Numerical comparison of two ice crystal formation mechanisms on snowfall enhancement from ground-based aerosol generators. J. Appl. Meteor., 36, 70–85, https://doi.org/10.1175/1520-0450(1997)036<0070: NCOTIC>2.0.CO;2.
- Lin, Y., and B. A. Colle, 2009: The 4–5 December 2001 IMPROVE-2 event: Observed microphysics and comparisons with the Weather Research and Forecasting Model. *Mon. Wea. Rev.*, 137, 1372–1392, https://doi.org/10.1175/2008MWR2653.1.
- Liu, C., and Coauthors, 2017: Continental-scale convectionpermitting modeling of the current and future climate of North America. *Climate Dyn.*, 49, 71–95, https://doi.org/10.1007/s00382-016-3327-9.
- Long, A. B., 2001: Review of persistence effects of silver iodide seeding. J. Wea. Modif., 33, 9–23.
- —, and A. W. Huggins, 1992: Australian Winter Storms Experiment (AWSE) I: Supercooled liquid water and precipitationenhancement opportunities. J. Appl. Meteor., 31, 1041–1055, https://doi.org/10.1175/1520-0450(1992)031<1041:AWSEIS> 2.0.CO;2.
- , and E. J. Carter, 1996: Australian winter mountain storm clouds: Precipitation augmentation potential. *J. Appl. Meteor.*, 35, 1457–1464, https://doi.org/10.1175/1520-0450(1996)035<1457: AWMSCP>2.0.CO;2.
- Manton, M. J., and L. Warren, 2011: A confirmatory snowfall enhancement project in the Snowy Mountains of Australia. Part II: Primary and associated analyses. J. Appl. Meteor. Climatol., 50, 1448–1458, https://doi.org/10.1175/2011JAMC2660.1.
 - —, —, S. L. Kenyon, A. D. Peace, S. P. Bilish, and K. Kemsley, 2011: A confirmatory snowfall enhancement project in the Snowy Mountains of Australia. Part I: Project design and response variables. J. Appl. Meteor. Climatol., 50, 1432–1447, https://doi.org/10.1175/2011JAMC2659.1.
- —, A. D. Peace, K. Kemsley, S. Kenyon, J. C. Speirs, L. Warren, and J. Denholm, 2017: Further analysis of a snowfall enhancement project in the Snowy Mountains of Australia. *Atmos. Res.*, **193**, 192–203, https://doi.org/10.1016/j.atmosres.2017.04.011.
- Marwitz, J. D., 1987: Deep orographic storms over the Sierra Nevada. Part II: The precipitation processes. *J. Atmos. Sci.*, **44**, 174–185, https://doi.org/10.1175/1520-0469(1987)044<0174: DOSOTS>2.0.CO;2.
- —, and R. E. Stewart, 1981: Some seeding signatures in Sierra storms. J. Appl. Meteor., 20, 1129–1144, https://doi.org/ 10.1175/1520-0450(1981)020<1129:SSSISS>2.0.CO;2.
- McFarquhar, G. M., D. Baumgardner, and A. J. Heymsfield, 2017: Background and overview. *Ice Formation and Evolution in Clouds and Precipitation: Measurement and Modeling Challenges, Meteor. Monogr.*, No. 58, v–ix, Amer. Meteor. Soc, https://doi.org/10.1175/AMSMONOGRAPHS-D-16-0018.1.
- Medina, S., and R. A. Houze, 2015: Small-scale precipitation elements in midlatitude cyclones crossing the California Sierra Nevada. *Mon. Wea. Rev.*, **143**, 2842–2870, https://doi.org/ 10.1175/MWR-D-14-00124.1.

2136

- Meyers, M. P., P. J. Demott, and W. R. Cotton, 1995: A comparison of seeded and nonseeded orographic cloud simulations with an explicit cloud model. J. Appl. Meteor., 34, 834–846, https://doi.org/10.1175/1520-0450(1995)034<0834: ACOSAN>2.0.CO;2.
- Mielke, P. W., 1995: Comments on the Climax I and II Experiments including replies to Rangno and Hobbs. J. Appl. Meteor., 34, 1228–1232, https://doi.org/10.1175/1520-0450(1995)034<1228: COTCIA>2.0.CO;2.
- —, L. O. Grant, and C. F. Chappell, 1970: Elevation and spatial variation effects of wintertime orographic cloud seeding. *J. Appl. Meteor.*, 9, 476–488, https://doi.org/10.1175/1520-0450(1970)009<0476:EASVEO>2.0.CO;2.
 - —, —, and —, 1971: An independent replication of the Climax Wintertime Orographic Cloud Seeding Experiment. J. Appl. Meteor., 10, 1198–1212, https://doi.org/10.1175/1520-0450(1971)010<1198:AIROTC>2.0.CO;2.
- —, J. S. Williams, and S. Wu, 1977: Covariance analysis technique based on bivariate log-normal distribution with weather modification applications. J. Appl. Meteor., 16, 183–187, https://doi.org/ 10.1175/1520-0450(1977)016<0183:CATBOB>2.0.CO;2.
- —, G. W. Brier, L. O. Grant, G. J. Mulvey, and P. N. Rosenzweig, 1981: A Statistical reanalysis of the replicated Climax I and II wintertime orographic cloud seeding experiments. J. Appl. Meteor., 20, 643–659, https://doi.org/10.1175/ 1520-0450(1981)020<0643:ASROTR>2.0.CO;2.
- Minder, J. R., D. R. Durran, G.H. Roe, and A. M. Anders, 2008: The climatology of small-scale orographic precipitation over the Olympic Mountains: Patterns and processes. *Quart. J. Roy. Meteor. Soc.*, **134**, 817–839, https://doi.org/10.1002/qj.258.
- Morrison, A. E., S. T. Siems, M. J. Manton, and A. Nazarov, 2009: On the analysis of a cloud seeding dataset over Tasmania. J. Appl. Meteor. Climatol., 48, 1267–1280, https://doi.org/ 10.1175/2008JAMC2068.1.
 - —, —, and —, 2013: On a natural environment for glaciogenic cloud seeding. J. Appl. Meteor. Climatol., 52, 1097–1104, https://doi.org/10.1175/JAMC-D-12-0108.1.
- Morrison, H., and W. W. Grabowski, 2008: A novel approach for representing ice microphysics in models: Description and tests using a kinematic framework. J. Atmos. Sci., 65, 1528–1548, https://doi.org/10.1175/2007JAS2491.1.
- Mote, P. W., A. F. Hamlet, M. P. Clark, and D. P. Lettenmaier, 2005: Declining mountain snowpack in western North America. *Bull. Amer. Meteor. Soc.*, 86, 39–50, https://doi.org/10.1175/ BAMS-86-1-39.
- NRC, 2003: Critical Issues in Weather Modification Research. National Academies Press, 123 pp.
- Neyman, J., E. L. Scott, and M. Vasilevskis, 1960: Statistical evaluation of the Santa Barbara randomized cloud seeding experiment. *Bull. Amer. Meteor. Soc.*, **41**, 531–547, https:// doi.org/10.1175/1520-0477-41.10.531.
- Orville, H. D., 1996: A review of cloud modeling in weather modification. *Bull. Amer. Meteor. Soc.*, **77**, 1535–1555, https://doi.org/ 10.1175/1520-0477(1996)077<1535:AROCMI>2.0.CO;2.
- Plooster, M. N., and N. Fukuta, 1975: A numerical model of precipitation from seeded and unseeded cold orographic clouds. *J. Appl. Meteor.*, 14, 859–867, https://doi.org/10.1175/1520-0450(1975)014<0859:ANMOPF>2.0.CO;2.
- Plummer, D. M., G. M. McFarquhar, R. M. Rauber, B. F. Jewett, and D. C. Leon, 2014: Structure and statistical analysis of the microphysical properties of generating cells in the commahead region of continental winter cyclones. J. Atmos. Sci., 71, 4181–4203, https://doi.org/10.1175/JAS-D-14-0100.1.

- Pokharel, B., and B. Geerts, 2016: A multi-sensor study of the impact of ground-based glaciogenic seeding on clouds and precipitation over mountains in Wyoming. Part I: Project description. Atmos. Res., 182, 269–281, https://doi.org/10.1016/ j.atmosres.2016.08.008.
 - —, —, and X. Jing, 2014a: The impact of ground-based glaciogenic seeding on orographic clouds and precipitation: a multi-sensor case study. J. Appl. Meteor. Climatol., 53, 890– 909, https://doi.org/10.1175/JAMC-D-13-0290.1.
 - —, —, K. Friedrich, J. Aikins, D. Breed, R. Rasmussen, and A. Huggins, 2014b: The impact of ground-based glaciogenic seeding on clouds and precipitation over mountains: A multi-sensor case study of shallow precipitating orographic cumuli. *Atmos. Res.*, **147–148**, 162–182, https://doi.org/ 10.1016/j.atmosres.2014.05.014.
 - —, —, and —, 2015: The impact of ground-based glaciogenic seeding on clouds and precipitation over mountains: A case study of a shallow orographic cloud with large supercooled droplets. J. Geophys. Res. Atmos., **120**, 6056–6079, https:// doi.org/10.1002/2014JD022693.
 - —, —, K. Friedrich, K. Ikeda, and R. Rasmussen, 2017: A multi-sensor study of the impact of ground-based glaciogenic seeding on clouds and precipitation over mountains in Wyoming. Part II: Seeding impact analysis. *Atmos. Res.*, **183**, 42–57, https://doi.org/10.1016/j.atmosres.2016.08.018.
- —, —, and —, 2018: The impact of ground-based glaciogenic seeding on a shallow stratiform cloud over the Sierra Madre in Wyoming: A multi-sensor study of the 3 March 2012 case. *Atmos. Res.*, **214**, 74–90, https://doi.org/10.1016/j.atmosres.2018.07.013.
- Politovich, M. K., and G. Vali, 1983: Observations of liquid water in orographic clouds over Elk Mountain. J. Atmos. Sci., 40, 1300–1312, https://doi.org/10.1175/1520-0469(1983)040<1300: OOLWIO>2.0.CO;2.
- Prasad, N., A. R. Rodi, and A. J. Heymsfield, 1989: Observations and numerical simulations of precipitation development in seeded clouds over the Sierra Nevada. J. Appl. Meteor., 28, 1031–1049, https://doi.org/10.1175/1520-0450(1989)028<1031: OANSOP>2.0.CO:2.
- Pruppacher, H. R., and J. D. Klett, 2010: Microphysics of Clouds and Precipitation. Springer, 852 pp.
- Rangno, A. L., 1979: A reanalysis of the Wolf Creek Pass cloud seeding experiment. J. Appl. Meteor., 18, 579–605, https://doi.org/ 10.1175/1520-0450(1979)018<0579:AROTWC>2.0.CO;2.
- —, 1986: How good are our conceptual models of orographic cloud seeding? *Precipitation Enhancement*—A Scientific Challenge, Meteor. Monogr., No. 43, Amer. Meteor. Soc., 115– 126, https://doi.org/10.1175/0065-9401-21.43.115.
- —, and P. V. Hobbs, 1980a: Comments on "Generalized criteria for seeding winter orographic clouds." *J. Appl. Meteor.*, **19**, 906–907, https://doi.org/10.1175/1520-0450(1980)019<0906: COCFSW>2.0.CO;2.
- —, and —, 1980b: Comments on "Randomized cloud seeding in the San Juan Mountains, Colorado." J. Appl. Meteor., 19, 346–350, https://doi.org/10.1175/1520-0450(1980)019<0346: COCSIT>2.0.CO;2.
- —, and —, 1981: Comments on "Reanalysis of 'Generalized criteria for seeding winter orographic clouds'." J. Appl. Meteor., 20, 216–217, https://doi.org/10.1175/1520-0450(1981) 020<0216:COOCFS>2.0.CO;2.
- —, and —, 1987: A reevaluation of the Climax cloud seeding experiments using NOAA published data. J. Climate Appl. Meteor., 26, 757–762, https://doi.org/10.1175/1520-0450(1987) 026<0757:AROTCC>2.0.CO;2.

—, and —, 1993: Further analyses of the Climax cloud-seeding experiments. J. Appl. Meteor., 32, 1837–1847, https://doi.org/ 10.1175/1520-0450(1993)032<1837:FAOTCC>2.0.CO;2.

- Rasmussen, R., and Coauthors, 2011: High-resolution coupled climate runoff simulations of seasonal snowfall over Colorado: A process study of current and warmer climate. *J. Climate*, 24, 3015–3048, https://doi.org/10.1175/2010JCLI3985.1.
- —, and Coauthors, 2012: How well are we measuring snow? The NOAA/FAA/NCAR Winter Precipitation Test Bed. Bull. Amer. Meteor. Soc., 93, 811–829, https://doi.org/10.1175/ BAMS-D-11-00052.1.
- Rasmussen, R. M., and Coauthors, 2018: Evaluation of the Wyoming Weather Modification Pilot Project (WWMPP) using two approaches: Traditional statistics and ensemble modeling. J. Appl. Meteor. Climatol., 57, 2639–2660, https:// doi.org/10.1175/JAMC-D-17-0335.1.
- Rauber, R. M., 1987: Characteristics of cloud ice and precipitation during wintertime storms over the mountains of northern Colorado. J. Climate Appl. Meteor., 26, 488–524, https://doi.org/ 10.1175/1520-0450(1987)026<0488:COCIAP>2.0.CO;2.
- —, 1992: Microphysical structure and evolution of a central Sierra Nevada orographic cloud system. J. Appl. Meteor., 31, 3–24, https://doi.org/10.1175/1520-0450(1992)031<0003: MSAEOA>2.0.CO;2.
- —, and L. O. Grant, 1986: The characteristics and distribution of cloud water over the mountains of northern Colorado during wintertime storms. Part II: Spatial distribution and microphysical characteristics. J. Climate Appl. Meteor., 25, 489–504, https://doi.org/10.1175/1520-0450(1986)025<0489: TCADOC>2.0.CO;2.
- —, and —, 1987: Supercooled liquid water structure of a shallow orographic cloud system in southern Utah. J. Climate Appl. Meteor., 26, 208–215, https://doi.org/10.1175/1520-0450(1987)026<0208:SLWSOA>2.0.CO;2.
- —, and A. Tokay, 1991: An explanation for the existence of supercooled water at the top of cold clouds. J. Atmos. Sci., 48, 1005–1023, https://doi.org/10.1175/1520-0469(1991)048<1005: AEFTEO>2.0.CO;2.
- —, L. O. Grant, D.-X. Feng, and J. B. Snider, 1986: The characteristics and distribution of cloud water over the mountains of northern Colorado during wintertime storms. Part I: Temporal variations. J. Climate Appl. Meteor., 25, 468–488, https:// doi.org/10.1175/1520-0450(1986)025<0468:TCADOC>2.0.CO;2.
- —, R. D. Elliott, J. O. Rhea, A. W. Huggins, and D. W. Reynolds, 1988: A diagnostic technique for targeting during airborne seeding experiments in wintertime storms over the Sierra Nevada. J. Appl. Meteor., 27, 811–828, https://doi.org/ 10.1175/1520-0450(1988)027<0811:ADTFTD>2.0.CO;2.
- Reinking, R. F., 1979: The onset and early growth of snow crystals by accretion of droplets. *J. Atmos. Sci.*, **36**, 870–881, https://doi.org/ 10.1175/1520-0469(1979)036<0870:TOAEGO>2.0.CO;2.
- —, J. B. Snider, and J. L. Coen, 2000: Influences of stormembedded orographic gravity waves on cloud liquid water and precipitation. J. Appl. Meteor., 39, 733–759, https://doi.org/ 10.1175/1520-0450(2000)039<0733:IOSEOG>2.0.CO;2.
- Reisin, T., Z. Levin, and S. Tzivion, 1996: Rain production in convective clouds as simulated in an axisymmetric model with detailed microphysics. Part I: Description of the model. *J. Atmos. Sci.*, **53**, 497–519, https://doi.org/10.1175/1520-0469(1996)053<0497:RPICCA>2.0.CO;2.
- Reynolds, D. W., 1988: A report on winter snowpack-augmentation. Bull. Amer. Meteor. Soc., 69, 1290–1300, https://doi.org/ 10.1175/1520-0477(1988)069<1290:AROWSA>2.0.CO;2.

- —, 2015: Literature review and scientific synthesis on the efficacy of winter orographic cloud seeding: A report to the U.S. Bureau of Reclamation. CIRES Tech. Memo. Statement on the Application of Winter Orographic Cloud Seeding For Water Supply and Energy Production, 148 pp., https://www.usbr.gov/ main/qoi/docs/Literature_Review_and_Scientific_Synthesis_ of_the_Efficacy_of_Winter_Orographic_Cloud_Seeding_Peer_ Review.pdf.
- —, and A. S. Dennis, 1986: A review of the Sierra Cooperative Pilot Project. Bull. Amer. Meteor. Soc., 67, 513–523, https:// doi.org/10.1175/1520-0477(1986)067<0513:AROTSC>2.0.CO;2.
- Ritzman, J., T. Deshler, K. Ikeda, and R. Rasmussen, 2015: Estimating the fraction of winter orographic precipitation produced under conditions meeting the seeding criteria for the Wyoming Weather Modification Pilot Project. J. Appl. Meteor. Climatol., 54, 1202–1215, https://doi.org/10.1175/JAMC-D-14-0163.1.
- Rosenow, A., R. M. Rauber, G. M. McFarquhar, B. F. Jewett, D. Plummer, and D. Leon, 2014: Vertical velocity and physical structure of generating cells and elevated convection in the comma-head region of continental of winter cyclones. *J. Atmos. Sci.*, **71**, 1538–1558, https://doi.org/10.1175/JAS-D-13-0249.1.
- Rottner, D., L. Vardiman, and J. A. Moore, 1980: Reanalysis of "Generalized criteria for seeding winter orographic clouds." *J. Appl. Meteor.*, **19**, 622–626, https://doi.org/10.1175/1520-0450(1980)019<0622:ROCFSW>2.0.CO;2.
- Ryan, B. F., and W. D. King, 1997: A critical review of the Australian experience in cloud seeding. *Bull. Amer. Meteor. Soc.*, 78, 239–254, https://doi.org/10.1175/1520-0477(1997)078<0239: ACROTA>2.0.CO;2.
- Saleeby, S. M., W. R. Cotton, D. Lowenthal, R. D. Borys, and M. A. Wetzel, 2009: Influence of cloud condensation nuclei on orographic snowfall. J. Appl. Meteor. Climatol., 48, 903–922, https://doi.org/10.1175/2008JAMC1989.1.
- —, —, and J. D. Fuller, 2011: The cumulative impact of cloud droplet nucleating aerosols on orographic snowfall in Colorado. J. Appl. Meteor. Climatol., 50, 604–625, https://doi.org/10.1175/ 2010JAMC2594.1.
- —, —, D. Lowenthal, and J. Messina, 2013: Aerosol impacts on the microphysical growth processes of orographic snowfall. J. Appl. Meteor. Climatol., 52, 834–852, https://doi.org/ 10.1175/JAMC-D-12-0193.1.
- Sarkadi, N., I. Geresdi, and G. Thompson, 2016: Numerical simulation of precipitation formation in the case orographically induced convective cloud: Comparison of the results of bin and bulk microphysical schemes. *Atmos. Res.*, 180, 241–261, https://doi.org/10.1016/j.atmosres.2016.04.010.
- Sassen, K., and H. Zhao, 1993: Supercooled liquid water clouds in Utah winter mountain storms: Cloud-seeding implications of a remote-sensing dataset. J. Appl. Meteor., 32, 1548–1558, https:// doi.org/10.1175/1520-0450(1993)032<1548:SLWCIU>2.0.CO;2.
- —, R. M. Rauber, and J. B. Snider, 1986: Multiple remote sensor observations of supercooled liquid water in a winter storm at Beaver, Utah. J. Climate Appl. Meteor., 25, 825–834, https://doi.org/10.1175/1520-0450(1986)025<0825: MRSOOS>2.0.CO;2.
- —, A. W. Huggins, A. B. Long, J. B. Snider, and R. J. Meitín, 1990: Investigations of a winter mountain storm in Utah. Part II: Mesoscale structure, supercooled liquid water development, and precipitation processes. J. Atmos. Sci., 47, 1323–1350, https://doi.org/10.1175/1520-0469(1990)047<1323: IOAWMS>2.0.CO;2.

- Schaefer, V. J., 1946: The production of ice crystals in a cloud of supercooled water droplets. *Science*, **104**, 457–459, https:// doi.org/10.1126/science.104.2707.457.
- Silverman, B. A., 2001: A critical assessment of glaciogenic seeding of convective clouds for rainfall enhancement. *Bull. Amer. Meteor. Soc.*, **82**, 903–923, https://doi.org/10.1175/1520-0477(2001)082<0903:ACAOGS>2.3.CO;2.
- —, 2007: On the use of ratio statistics for the evaluation of operational cloud seeding programs. J. Wea. Modif., 39, 50–60.
- —, 2008: A statistical evaluation of the Kern River Operational Cloud Seeding Program. J. Wea. Modif., 40, 7–16.
- —, 2009: An evaluation of the San Joaquin operational cloud seeding program using Monte Carlo permutation statistics. *J. Wea. Modif.*, **41**, 15–22.
- Smith, R. B., 1979: The influence of mountains on the atmosphere. Advances in Geophysics, Vol. 21, Academic Press, 82–230, https://doi.org/10.1016/S0065-2687(08)60262-9.
- —, and I. Barstad, 2004: A linear theory of orographic precipitation. J. Atmos. Sci., 61, 1377–1391, https://doi.org/ 10.1175/1520-0469(2004)061<1377:ALTOOP>2.0.CO;2.
- Stein, A. F., R. R. Draxler, G. D. Rolph, B. J. Stunder, M. D. Cohen, and F. Ngan, 2015: NOAA's HYSPLIT atmospheric transport and dispersion modeling system. *Bull. Amer. Meteor. Soc.*, 96, 2059–2077, https://doi.org/10.1175/BAMS-D-14-00110.1.
- Stoelinga, M. T., and Coauthors, 2003: Improvement of microphysical parameterization through observational verification experiment. *Bull. Amer. Meteor. Soc.*, 84, 1807–1826, https:// doi.org/10.1175/BAMS-84-12-1807.
- Super, A. B., 1974: Silver iodide plume characteristics over the Bridger Mountain Range, Montana. J. Appl. Meteor., 13, 62–70, https://doi.org/10.1175/1520-0450(1974)013<0062: SIPCOT>2.0.CO;2.
- —, 1986: Further exploratory analysis of the Bridger Range Winter Cloud Seeding Experiment. *J. Climate Appl. Meteor.*, 25, 1926–1933, https://doi.org/10.1175/1520-0450(1986)025<1926: FEAOTB>2.0.CO;2.
- —, and J. A. Heimbach, 1983: Evaluation of the Bridger Range Winter Cloud Seeding Experiment using control gages. J. Climate Appl. Meteor., 22, 1989–2011, https://doi.org/ 10.1175/1520-0450(1983)022<1989:EOTBRW>2.0.CO;2.
- —, and B. A. Boe, 1988: Microphysical effects of wintertime cloud seeding with silver iodide over the Rocky Mountains. Part III: Observations over the Grand Mesa, Colorado. *J. Appl. Meteor.*, **27**, 1166–1182, https://doi.org/10.1175/1520-0450(1988)027<1166:MEOWCS>2.0.CO;2.
- —, and J. A. Heimbach, 1988: Microphysical effects of wintertime cloud seeding with silver iodide over the Rocky Mountains. Part II: Observations over the Bridger Range, Montana. *J. Appl. Meteor.*, **27**, 1152–1165, https://doi.org/10.1175/1520-0450(1988)027<1152:MEOWCS>2.0.CO;2.
- —, and —, 2003: Reexamination of historical regression analysis applied to a recent Idaho cloud seeding project. *J. Wea. Modif.*, **35**, 25–40.
- —, and —, 2009: Six hour analyses of the Bridger Range Randomized Winter Orographic Cloud Seeding Experiment. J. Wea. Modif., 41, 38–58.
- Sykes, R. I., and R. S. Gabruk, 1997: A second-order closure model for the effect of averaging time on turbulent plume dispersion. *J. Appl. Meteor.*, **36**, 1038–1045, https://doi.org/10.1175/1520-0450(1997)036<1038:ASOCMF>2.0.CO;2.
- Tessendorf, S. A., B. Boe, B. Geerts, M. J. Manton, S. Parkinson, and R. Rasmussen, 2015: The future of winter orographic

cloud seeding: A view from scientists and stakeholders. *Bull. Amer. Meteor. Soc.*, **96**, 2195–2198, https://doi.org/10.1175/ BAMS-D-15-00146.1.

- —, and Coauthors, 2017: Developing improved products to forecast and diagnose aircraft icing conditions based upon drop size. AIAA Aviation Forum Proc., Denver, CO, AIAA, AIAA 2017-4473, https://doi.org/10.2514/6.2017-4473.
- —, and Coauthors, 2019: Transformational approach to winter orographic weather modification research: The SNOWIE Project. *Bull. Amer. Meteor. Soc.*, **100**, 71–92, https://doi.org/ 10.1175/BAMS-D-17-0152.1.
- Thompson, G., and T. Eidhammer, 2014: A study of aerosol impacts on clouds and precipitation development in a large winter cyclone. J. Atmos. Sci., 71, 3636–3658, https://doi.org/ 10.1175/JAS-D-13-0305.1.
- —, P. R. Field, W. R. Hall, and R. Rasmussen, 2008: Explicit forecasts of winter precipitation using an improved bulk microphysics scheme. Part II: Implementation of a new snow parameterization. *Mon. Wea. Rev.*, **136**, 5095–5115, https:// doi.org/10.1175/2008MWR2387.1.
- U.S. Census Bureau, 2010: 2010 Census: Apportionment data. U.S. Census Bureau, https://www.census.gov/data/tables/2010/dec/apportionment-data-text.html.
- Uttal, T., R. M. Rauber, and L. O. Grant, 1988: Distributions of liquid, vapor, and ice in an orographic cloud from field observations. J. Atmos. Sci., 45, 1110–1122, https://doi.org/ 10.1175/1520-0469(1988)045<1110:DOLVAI>2.0.CO:2.
- Vali, G., D. Leon, and J. R. Snider, 2012: Ground-layer snow clouds. *Quart. J. Roy. Meteor. Soc.*, **138**, 1507–1525, https:// doi.org/10.1002/qj.1882.
- Vardiman, L., and J. A. Moore, 1978: Generalized criteria for seeding winter orographic clouds. J. Appl. Meteor., 17, 1769– 1777, https://doi.org/10.1175/1520-0450(1978)017<1769: GCFSWO>2.0.CO;2.
- Vonnegut, B., 1947: The nucleation of ice formation by silver iodide. J. Appl. Phys., 18, 593–595, https://doi.org/10.1063/ 1.1697813.
- Warburton, J. A., L. G. Young, and R. H. Stone, 1995: Assessment of seeding effects in snowpack augmentation programs: Ice nucleation and scavenging of seeding aerosols. J. Appl. Meteor., 34, 121–130, https://doi.org/10.1175/1520-0450-34.1.121.
- Weather Modification, Inc., 2005: Wyoming Level II Weather Modification Feasibility Study. Wyoming Water Development Commission Final Rep., 151 pp., http://library.wrds.uwyo.edu/ wwdcrept/Wyoming/Wyoming-Level_II_Weather_Modification_ Feasibility_Study-Final_Report-2005.html.
- Wegener, A., 1911: Thermodynamik der Atmosphäre (Thermodynamics of the Atmosphere). J. A. Barth, 331 pp.
- Weigel, A. P., F. K. Chow, and M. W. Rotach, 2007: On the nature of turbulent kinetic energy in a steep and narrow Alpine valley. *Bound.-Layer Meteor.*, **123**, 177–199, https://doi.org/ 10.1007/s10546-006-9142-9.
- Wrzesien, M. L., M. T. Durand, and T. M. Pavelsky, 2019: A reassessment of North American river basin cool-season precipitation: Developments from a new mountain climatology data set. *Water Resour. Res.*, 55, 3502–3519, https://doi.org/ 10.1029/2018WR024106.
- Xue, L., and Coauthors, 2013a: Implementation of a silver iodide cloud seeding parameterization in WRF. Part I: Model description and idealized 2D sensitivity tests. J. Appl. Meteor. Climatol., 52, 1433–1457, https://doi.org/10.1175/JAMC-D-12-0148.1.
- —, S. Tessendorf, E. Nelson, R. Rasmussen, D. Breed, S. Parkinson, P. Holbrook, and D. Blestrud, 2013b: Implementation of a silver

iodide cloud seeding parameterization in WRF. Part II: 3D simulations of actual seeding events and sensitivity tests. *J. Appl. Meteor. Climatol.*, **52**, 1458–1476, https://doi.org/10.1175/JAMC-D-12-0149.1.

- —, X. Chu, R. Rasmussen, D. Breed, B. Boe, and B. Geerts, 2014: The dispersion of silver iodide particles from groundbased generators over complex terrain. Part II: WRF large-eddy simulations versus observations. J. Appl. Meteor. Climatol., 53, 1342–1361, https://doi.org/10.1175/JAMC-D-13-0241.1.
 - —, —, —, , and B. Geerts, 2016: A case study of radar observations and WRF LES simulations of the impact of ground-based glaciogenic seeding on orographic clouds and precipitation. Part II: AgI dispersion and seeding signals simulated by WRF. J. Appl. Meteor. Climatol., 55, 445–464, https://doi.org/10.1175/JAMC-D-15-0115.1.
- —, and Coauthors, 2017: WRF large-eddy simulations of chemical tracer deposition and seeding effect over complex terrain from ground- and aircraft-based AgI generators. *Atmos. Res.*, **190**, 89–103, https://doi.org/10.1016/ j.atmosres.2017.02.013.
- Yin, Y., Z. Levin, T. G. Reisin, and S. Tzivion, 2000: Seeding convective clouds with hygroscopic flares: Numerical simulations using a cloud model with detailed microphysics. *J. Appl. Meteor.*, **39**, 1460–1472, https://doi.org/10.1175/1520-0450(2000)039<1460:SCCWHF>2.0.CO;2.
- Young, K. C., 1974: A numerical simulation of wintertime, orographic precipitation: Part II. Comparison of natural and AgI-seeded conditions. J. Atmos. Sci., 31, 1749–1767, https://doi.org/10.1175/1520-0469(1974)031<1749:ANSOWO> 2.0,CO;2.